

GOPEN ACCESS

Citation: Chen Q, Yang X, Yang D, Liu S, Liao C (2024) Blockchain technology for supply chain traceability: A game-theoretic analysis between e-platforms. PLoS ONE 19(4): e0297978. https://doi.org/10.1371/journal.pone.0297978

Editor: Amir M. Fathollahi-Fard, University of Victoria / Universiti Teknologi Malaysia /, CANADA

Received: May 29, 2023

Accepted: January 15, 2024

Published: April 25, 2024

Copyright: © 2024 Chen et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript.

Funding: This work was supported by the National Natural Science Foundation of China [grant number 71862035]; the Yunnan Provincial Science and Technology Research Project of China National Tobacco Corporation [grant number 2023530000241031]; and The 21st Yunnan Young and Middle-aged Academic and Technical Leaders Reserve Personnel Training Program [grant number 2019HB030]. **RESEARCH ARTICLE**

Blockchain technology for supply chain traceability: A game-theoretic analysis between e-platforms

Qian Chen¹°, Xuejian Yang²°, Dan Yang³, Sen Liu⁶, ^{4,5}*, Chuchu Liao⁴

 School of Management, Hefei University of Technology, Hefei, China, 2 Dali Tobacco Company of Yunnan Province, Dali, China, 3 International College, Dhurakij Pundit University, Bangkok, Thailand, 4 School of Logistics and Management Engineering, Yunnan University of Finance and Economics, Kunming, China, 5 Yunnan Key Laboratory of Service Computing, Kunming, China

So These authors contributed equally to this work.

* liusencool@ynufe.edu.cn

Abstract

In view of the rapid development and application of blockchain technology, this paper considers a secondary supply chain system consisting of a single upstream supplier and a downstream e-tailer that implements blockchain technology and a traditional e-tailer that does not implement blockchain technology. We establish the demand function of two channels based on consumers' sensitivity to the blockchain and use the Stackelberg game model to compare and analyse the e-tailers' profits from the two channels. For the basic properties, interestingly, we find that there exists a critical threshold on the cost sensitivity to effort that helps each e-tailer decide whether to implement effort. If the cost sensitivity to effort is high, the two-sided e-tailers will reduce their effort as much as possible to obtain greater profits. Conversely, if the cost sensitivity to effort is low, they will increase their effort to obtain more benefits. We also discuss the role of blockchain technology in competition between e-tailers and analyse the impact of the product brand effect brought by the traceability characteristic of blockchain on the competition between e-tailers. To check the robustness of the core results and to investigate different forms of supply chain configurations, this paper further develops the analysis of the supplier entering agency agreements with two-sided e-tailers. Under this scenario, the supplier sells their products directly to consumers through two-sided e-platforms and shares revenue with e-tailers as platform agency fees. These core ideas remain valid in the extended model.

1. Introduction

Dual-channel competition has always been the focus of research in the field of supply chains, but the subjects of competition are not the same. In recent years, due to the development and application of blockchain technology [1], as well as consumers' demands for product authenticity [2, 3], some retailers have been prompted to adopt blockchain technology to provide consumers with product source traceability services, including Walmart and JD. Although the

Competing interests: The authors have declared that no competing interests exist.

application of this technology is mostly on a small scale, it still has a certain impact on traditional retail platforms [4]. As a result, a new competitive model is formed in the market: horizontal competition between a new retail platform that provides blockchain technology and a traditional retail platform that does not use blockchain technology.

In the retail industry, lack of supply chain traceability is a challenge to check that products are genuine and safe. For this pain point, blockchain technology has the characteristics of traceability; it helps to improve the safety and authenticity of products and solve the consumer's trust crisis in the product [4–6]. For example, the product traceability service brought by blockchain technology has had a positive spillover effect on Walmart's brand [7]. Although related research has found that the traceability of blockchain can indeed bring consumer stickiness, which we call the brand value effect (JD.com, 2020), the traceability system will also bring costs per unit of product, such as product information entry and information maintenance on the chain [8, 9]. Therefore, traditional retail platforms that do not use blockchain technology can still compete with blockchain retail platforms through other marketing efforts. In summary, motivated by the observed real-world industrial practice on e-platform operations for product sales services and the importance of product traceability in the blockchain era, this paper will study the horizontal competition between blockchain platforms and traditional platforms.

Although horizontal competition for e-commerce exists and there is an urgency to address the aforementioned issues, previous literature has focused on investigating the inner mechanism within a particular blockchain platform [10, 11]. The horizontal competition between etailers in a dual-channel supply chain based on blockchain technology remains largely unexplored. In this paper, we aim to fill this research gap by analysing the competition between electronic platforms from the perspective of the brand value effect brought by the traceability of the blockchain.

Specifically, we address the following research questions.

- What factors affect the profitability of dual-channel supply chain members?
- What factors affect the competition between the two e-platforms?
- To check the robustness of the core results, is it possible to obtain similar results in the extended model?

To answer these questions and fill the void in the literature, we develop a dual-channel platform competition game-theoretic duopoly economic model based on the heterogeneity of consumers' sensitivity to blockchain technology, in which the two electronic platforms compete in the scenarios of using blockchain technology (the new e-tailer) and not using blockchain technology (the traditional e-tailer). Under both scenarios, competing e-tailers determine their platform effort based on the manufacturer's wholesale prices and ultimately obtain the optimal solution for the profits of dual-channel supply chain members. We provide numerical simulation to support our theoretical findings, enhance the practical implications, and provide management insights into electronic platform competition in the context of blockchain.

To the best of our knowledge, this paper is the first to establish a demand function based on consumers' sensitivity and the heterogeneity of blockchain technology. Meanwhile, this paper is also the first paper in the literature that analytically explores the product traceability between two competing e-platforms for product sales services in the application of blockchain technology. Specifically, our study shows the following insights. (i) For the basic properties, interest-ingly, we find that there exists a critical threshold on the cost sensitivity to effort that helps each e-tailer decide whether to implement effort. If the cost sensitivity to effort is high, two-sided e-tailers will reduce their effort as much as possible to obtain greater profits. Conversely,

if the cost sensitivity to effort is low, they will increase their effort to obtain more benefits. (ii) In the comparison of optimal profit, the results show that when the blockchain-sensitive consumer market has a relatively small market share but the brand value effect with blockchain is large, if the cost sensitivity to the effort is in a small range, the new e-tailer implementing blockchain obtains more benefits. However, when the blockchain-sensitive consumer market accounts for a relatively large market, no matter how the brand value effect with blockchain and the cost sensitivity to the effort are valued within their range, the profit level of the new etailer is always greater than that of the traditional e-tailer.

In addition, to check the robustness of the conclusions, we investigate the extended model by considering that the supplier enters agency selling agreements with e-platforms. In this scenario, the supplier can sell products directly to consumers through the blockchain e-platform and the traditional e-platform and share revenue with two-sided e-tailers as platform agency fees. Based on the model and numerical analysis, we find that the above core ideas remain valid in the extended model. Finally, the most important finding is that e-tailers are always more profitable in the agency model regardless of the scope of the cost sensitivity to effort. In particular, when the cost sensitivity to effort is low enough, the agency model may be a more efficient way to improve the benefits of supply chain members.

The remainder of this paper is organized as follows: The next section reviews the literature from three streams. Section 3 describes the model. Section 4 analyses the properties of decision variables and compares the profits of e-tailers. Section 5 verifies robustness of core results. Section 6 provides numerical experiments. Section 7 discusses the findings and provides managerial insights, and Section 8 is the conclusion.

2. Literature review

Our research is closely related to three streams of literature: literature on (i) blockchain technology in the supply chain, (ii) competition in the dual-channel supply chain, and (iii) heterogeneous consumers. We will consider the theme of heterogeneous consumers in a dualchannel supply chain competition based on the context of blockchain technology, and provide an overview of the third aspect above, following the logic of moving from surface to line to point. Next, we describe how our research relates to the literature in these areas.

2.1 Blockchain technology in supply chain

At present, the exploration of "blockchain + supply chain" is mainly presented in the literature around three major characteristics of the blockchain: the reduction of the bullwhip effect, smart contracts, and traceability. For example, Cole et al. [12] comprehensively and objectively considered the impact of various characteristics of the blockchain on the operation and management of the supply chain. They posit that whether to adopt blockchain technology is based on the characteristics of the supply chain. From the perspective of operations management, relevant scholars have tried to layout a research agenda for the OM field; they consider the impact of blockchain technology on classic operations management issues (such as the bullwhip effect, its causes and mitigation strategies) and further explore new business models to solve this problem, which is a focus of scholars and companies. With further research, related scholars have begun to quantitatively explore the influence of blockchain technology on the bullwhip effect. On the one hand, Giovanni [13] used game theory methods to compare the profit of supply chain members under the use of blockchain and traditional situations and finds that blockchain technology can reduce business risks and transaction costs. On the other hand, Chang et al. [14] clarified how the adoption of blockchain technology affects the best

ordering decisions and the corresponding best profits. The results show that the adoption of blockchain technology is not always profitable.

A smart contract is a computer protocol designed to facilitate, verify, or enforce the negotiation or execution of a contract. Specifically, the smart contract monitors the operations fulfilment as written in the blockchain, which allows us to confirm the planned process progress and to identify deviations. A smart contract typically stores rules and policies for negotiating terms and actions between parties. It automatically verifies that contractual terms have been met and executes transactions [15]. In view of this characteristic, Dolgui et al. [16] developed a model for blockchain-driven smart contract design and execution control and focus on the application of smart contracts in the field of "blockchain + supply chain". Moreover, through quantitative research, Giovanni [13] did find that smart revenue-sharing contracts are very efficient digital mechanisms and guarantee better performance than the sole blockchain.

The Hyperledger Fabric blockchain-based data module ensures the authenticity and traceability of the data source [17, 18]. The application of traceability can create transparency in supply chains [19]. With the specific company case, Bumblauskas et al. [20] aimed to illustrate the application of blockchain traceability technology in egg production and supply chain distribution systems from farms to consumers. Some scholars in related fields use qualitative methods to demonstrate the value of blockchain traceability to the supply chain in different contexts [21–23]; others use quantitative methods to compare and discover that the supply chain value brought by blockchain traceability is higher than that of the traditional supply chain [24, 25]. However, most of the articles found that whether blockchain traceability can always be beneficial requires supply chain members to weigh the relevant factors. Specifically, Zhang et al. explore how the grey market influences manufacturers' decisions to adopt blockchain by exploring whether and when manufacturers' adoption of blockchain discourages or facilitates grey marketers [26]. Xu et al. Consider a manufacturer's operational decisions in the presence of remanufacturing and blockchain [27]. Wu et al. focus on the optimal strategy for the adoption of blockchain technology in the supply chain of fresh produce, taking into account consumer preferences for traceability information [28]. Zhang et al. investigate the impact of different blockchain application scenarios on dual-channel supply chain decisions and further analyses the conditions for blockchain adoption [29]. Zhang et al. consider static versus dynamic pricing for blockchain technology platforms versus traditional platforms in the presence of network effects in a competitive environment [30]. However, different from the above research, Kouhizadeh et al. [31] discuss the application barriers of blockchain technology in the supply chain from the perspective of traceability. To show the difference more clearly, we summarize the previous literature regarding blockchain technology in supply chains. As shown in Table 1.

2.2 Competition in the dual-channel supply chain

Research on supply chains has been the focus of relevant scholars. These include supply chain risk management [32, 33], closed-loop supply chains [33–36], and dual-channel supply chains. However, this paper will focus on the dual-channel supply chain. There are many research points on the dual-channel supply chain. Among them, more attention is given to the horizon-tal competition of the two channels. Furthermore, competition between online and offline channels has always been the focus of scholars' research. Particularly noteworthy are the "showroom" and "BOPS" issues. Most research has found that the "showroom" intensifies the competition between traditional channels and electronic channels and reduces the profit of traditional retailers and e-tailers [37]. However, although some studies find that BOPS may intensify online and offline competition, they also discover that BOPS (Buy-Online-and-Pick-

Literature Research methods		The characteristics of block	chain		Involves dual-channel competition
		Reduce the bullwhip effect	Smart contracts	Transparent and traceability	
Cole et al. [12]	Qualitative	\checkmark	\checkmark	\checkmark	
Giovanni [13]	Quantitative	\checkmark	\checkmark		
Chang et al. [14]	Quantitative	\checkmark			
Saberi et al. [15]	Qualitative		\checkmark		
Dolgui et al. [16]	Quantitative		\checkmark		
Song et al. [17]	Quantitative			\checkmark	
Qiao et al. [18]	Quantitative			\checkmark	
Sunny et al. [19]	Qualitative				
Bumblauskas et al. [20]	Qualitative				
Mukherjee et al. [23]	Qualitative			\checkmark	
Pournader et al. [22]	Qualitative			\checkmark	
Hastig and Sodhi [21]	Qualitative			\checkmark	
Liu et al. [24]	Quantitative	\checkmark		\checkmark	
Liu et al. [25]	Quantitative			\checkmark	
Zhang et al. [26]	Quantitative			\checkmark	
Xu et al. [27]	Quantitative			\checkmark	
Wu et al. [28]	Quantitative			\checkmark	
Zhang et al. [29]	Quantitative			\checkmark	
Zhang et al. [30]	Quantitative				
Kouhizadeh et al. [31]	Qualitative				
Our study	Quantitative				2/

Table 1. Summary of the related literature regarding blockchain technology in supply chains.

Up-in-Store) revenue can be shared across channels to alleviate incentive conflicts [38]. In addition, Colombo and Matsushima [39] considered the spatial competition between offline and online retailers from the perspective of consumers' online and offline purchase efficiency. Chai et al. [40] showed that the store brand strategy may be an effective means for offline stores to ease the competition of e-commerce in opening exhibition halls.

Regarding the competition between online platforms, the issue of platform competition brought about by different sales models has always been a hot topic in the supply chain field. For example, Abhishek et al. [41] studied the competition between two e-tailers from the perspective of the choice of reselling model and agency selling model. Recently, some scholars have discussed the competition between two online platforms based on the perspective of blockchain technology. Choi et al. [42] established a duopoly game model of two rental service platforms with the support of blockchain technology and analyse the impact of information disclosure brought about by blockchain technology on platform competition. Different from the study of Choi, Jiang et al. [43] established a game theory model to study how the blockchain platform's decision on block size and transaction fees affects the competition between blockchain platforms. As shown in Table 2, we summarize the previous literature regarding the horizontal competition between two retailers in dual-channel supply chains.

2.3 Heterogeneous consumers

Consumer heterogeneity has always been a key issue for scholars in various fields, and many scholars build models based on different consumer heterogeneous types. Most scholars regard product price and quality as the most important factors that directly affect demand. Therefore,

Literature	Research Methodology	Competing members in the supply chain	Competition retailers	in dual-channel supply	ual-channel supply chains between	
			No involves	Involves brand value	Involves blockchain	
Moosavi et al. [<u>32</u>]	Literature Review					
Asghari [<u>35]</u>	Model analysis	Offline and offline				
Fathollahi et al. [<u>34</u>]	Design science					
Simonetto et al. [33]	Literature Review					
Berlin et al. [<u>36</u>]	Design science					
Balakrishnan et al. [37]	Model analysis	Online and offline	\checkmark			
Gao and Su [<u>38]</u>	Model analysis	Online and offline	\checkmark			
Colombo and Matsushima [39]	Model analysis	Online and offline	\checkmark			
Chai et al. [<u>40</u>]	Model analysis	Online and offline		\checkmark		
Abhishek et al. [41]	Model analysis	Online and online	\checkmark			
Choi et al. [<u>42</u>]	Model analysis	Online and online			\checkmark	
Jiang et al. [43]	Model analysis	Online and online				
Our study	Model analysis	Online and online			\checkmark	

Table 2. Summary of the related literature regarding the horizontal competition between two retailers in dual-channel supply chains.

they all use price and quality as indicators to distinguish consumer types to carry out research in the field of supply chain operations [44–48]. The focus of consumers' attention on products is ever changing. In addition to product price and quality, there are also consumers who pay more attention to product guarantees and product valuations. Therefore, Lei et al. [49] explore the issue of companies' dynamic pricing of products and warranty services based on the heterogeneity of consumers' attitudes towards warranty. Zheng et al. [50] and Qiu et al. [51] and Wang et al. [52] conduct research by examining the role of consumer valuation heterogeneity in corporate profits. Regarding food and green energy supply chain management, some researchers innovatively study the pricing of genetically modified foods by considering the heterogeneity of consumers' attitudes towards genetically modified foods [53, 54] study the implementation of green innovation in the entire supply chain by dividing heterogeneous consumers into two types: green and nongreen. Sharma et al. [55] consider heterogeneous consumer preferences for green and non-green luxury goods in a luxury context. Although there are related studies that use blockchain as a research background, the analysis is not based on consumers' sensitivity to blockchain. For example, Li et al. [56] discuss the pricing strategy of the supply chain from the perspective of consumers' heterogeneity of transaction costs in the context of blockchain. As shown in Table 3, we summarize the previous literature regarding the heterogeneity of consumers in supply chains.

2.4 Research gaps

By combing the above literature, we summarise the gaps between this paper and the relevant literature as follows:

First, the above studies are all theoretical and applied studies on today's popular technology, blockchain. Most articles reveal that the adoption of blockchain technology plays a major role in enhancing the competitiveness of enterprises in various industries. Similar to the above studies, we also discuss the impact of blockchain technology on supply chain operations. However, many articles use qualitative methods to analyse the benefits of blockchain characteristics for a single-channel supply chain. While some articles have quantitatively studied the impact of blockchain technology on the profitability of supply chain members through a gaming approach, very few articles have studied the impact of blockchain technology on competition

Literature	Consumer	Consumer heterogeneous type							
	Price or fee	Quality	Warranty	Genetic food	Green innovation	Product valuations	Blockchain		
Shi et al. [<u>44</u>]		\checkmark							
Herbon [45]		\checkmark							
Kopczewski et al. [46]	\checkmark								
Homburg et al. [47]	\checkmark								
Lin et al. [<u>48</u>]	\checkmark		\checkmark						
Lei et al. [49]			\checkmark						
Zheng et al. [50]						\checkmark			
Qiu et al. [51]						\checkmark			
Wang et al. [52]						\checkmark			
Li and Basu [53]				\checkmark					
Meng et al. [<u>54</u>]					\checkmark				
Sharma et al. [55]					\checkmark				
Li et al. [<u>56</u>]	\checkmark								
Our study							\checkmark		

Table 3. S	Summarv	of the re	elated lite	rature reg	arding	the heterog	reneitv	of consumers	in sup	plv d	chains
1 4010 01 0	,						,	01 00110411010		P-J '	

between two e-tailing platforms based on an e-commerce context. Therefore, based on the above studies, this paper focuses on competition between e-retailers in a dual-channel supply chain to further understand the value of blockchain adoption by discussing the relevance factors such as consumer sensitivity to blockchain, agency agreements, and brand value effects.

Second, the above articles have deeply studied the horizontal competition issues of dualchannel supply chain members. Similar to the above reviewed literature, this paper also focuses on horizontal competition between two retailers. However, the above literature studies fierce competition between online retailers and offline retailers. This paper mainly focuses on competition between two e-tailers. Although there are a few articles that explore the effect of brand value, these articles do not take blockchain technology into consideration. Moreover, they do not combine the blockchain with the brand value effect to study the dual-channel supply chain competition of two e-tailers. Therefore, this paper focuses on blockchain technology adoption decisions for competing platforms. Specifically, weighing the relationship between the brand spillover effects of blockchain technology and the cost of blockchain usage in an online dualchannel competitive environment is key to the analysis.

Third, the above studies are all based on the heterogeneity of consumers from different perspectives. Similar to the above reviewed literature, this paper also focuses on the heterogeneity of consumers. However, unlike all of them, this paper divides heterogeneous consumers into two types: blockchain-sensitive consumers and nonblockchain-sensitive consumers. Therefore, this paper builds a linear function model of demand based on the heterogeneity of consumer sensitivity to blockchain technology, which provides the research basis for this paper to conduct sensitivity analysis of relevant factors and profit comparison.

3. Model description and notations

To analyse the competition between electronic platforms from the perspective of the brand value effect brought by the traceability of the blockchain, we develop a dual-channel platform competition game-theoretic duopoly economic model based on the heterogeneity of consumers' sensitivity to blockchain technology, in which the two electronic platforms compete in the scenarios of using blockchain technology (the new e-tailer) and not using blockchain

Parameters	Definition
D^B	Demand in electronic channel with blockchain
D^E	Demand in electronic channel without blockchain
а	The market size
θ	The market share of blockchain-sensitive consumer
β	Cross-effort elasticity coefficient of demand
р	E-tail price
c _b	Cost in electronic channel with blockchain
C _e	Cost in electronic channel without blockchain
A	The cost sensitivity to effort
γ_b	Brand value effect with blockchain
Decision variables	Definition
W	Wholesale price of <i>S</i>
t _b	Effort with blockchain of <i>ER</i> ₁
t _e	Effort without blockchain of ER ₂

Table 4.	Parameter	definition.
----------	-----------	-------------

technology (the traditional e-tailer). We compare the equilibrium solutions in the two scenarios and analyse the factors affecting the profit level of competitive e-tailers. Moreover, we provide numerical simulation to support our theoretical findings, enhance the practical implications, and provide management insights into electronic platform competition in the context of blockchain technology.

3.1 Notations and definitions

To present a clearer picture of the meaning of the relevant parameters, the parameters needed in the modelling process are explained below in Table 4.

3.2 Model description

Consider a duopoly case in which in the market, two groups of e-tailers—e-tail channel with blockchain called new e-tailer (i.e., ER_1) and e-tail channel without blockchain called traditional e-tailer (i.e., ER_2), offer the same type of product to the market by directly selling the product from platforms which wholesale at a price *w* from the same supplier (i.e., *s*). In this paper, not surprisingly, we consider a two-level supply chain system consisting of a supplier and two competing e-tailers. Suppliers and two groups of e-tailers are risk neutral and completely rational; therefore, they make decisions according to the principle of expected profit maximization.

Some literature has classified heterogeneous consumers into two different types of consumers based on their different sensitivities to channels, and information sensitivities [44–47]. Similarly, we assume that there are two types of heterogeneous consumers in the market—a proportion of consumers θ prefer e-tail channels with blockchain (called "blockchain-sensitive"), and the rest 1– θ are blockchain-insensitive. Based on interviews with VeChain and Houbi China, Pun et al. [57] found that these two blockchain providers use a variable-cost model, and there is no fixed cost for using their blockchain service. Thus, according to the Pun's research [57], we assume that adopting blockchain technology incurs a unit variable-cost $c_b = \alpha t_b$ in the e-tail channel with blockchain. Similarly, a unit variable-cost $c_e = \alpha t_e$ for the e-tail channel without blockchain will be incurred, where cost is linearly related to the block-chain effort of ER_1 or the marketing effort of ER_2 (i.e., t_b or t_e) and hence increasing in t_b or t_e .



Fig 1. Supply chain structure under competition between the new e-tailer (blockchain technology-enabled) and the traditional e-tailer.

 α (0< α <1) is the cost sensitivity to the effort. Meanwhile, trust in the product is improved, and consumers are more loyal to the product when using blockchain technology, thus bringing consumer stickiness (JD.com, 2020). We call it the brand value effect $\gamma_b (\gamma_b \ge 0)$ and assume it an exogenous variable of the profit of the blockchain channel; otherwise, the platform will not adopt blockchain technology. It is worth noting that, through relevant literature research, we find that consumers who pay attention to the trust of product and brand, price is not their primary consideration (Marks and Spencer Press Release, 2018). In this article, we focus on the impact of the e-tailer's effort on whether it uses the blockchain (i.e., t_h and t_e) to improve consumer trust and thus affect demand. Therefore, we do not consider the impact of price on demand and assume that two e-tailers set the same retail price. As this paper focuses on the brand spillovers and unit adoption costs associated with the adoption of blockchain technology. Therefore, to simplify the research, this paper does not consider the initial investment cost of the implementation of the blockchain. Thus, we assume that the investment cost of the blockchain is normalized to zero. The supply chain structure diagram under competition between the new (blockchain technology-enabled) e-tailer and the traditional e-tailer in this paper is shown in Fig 1.

Specifically, Fig 1 shows that a single manufacturer first sells its products to a new e-tailer and a traditional e-tailer at wholesale price *w* respectively. For a new e-tailer with blockchain technology, it first adopts blockchain technology for its wholesale products and then sells them to consumers at retail price *p*. For a traditional e-tailer that does not adopt blockchain technology, it sells the products directly to consumers at retail price *p*. For traditional e-tailers that do not adopt blockchain technology, they sell their products directly to consumers at retail price *p*. We use a solid line to show the direction of product flow, along with a dashed line for the direction of payment flow. Specifically, for the product flow, products first flow from the supplier to both the new e-tailer and the traditional e-tailer at the same time, and then products with blockchain technology and products without blockchain technology flow from the new e-tailer and the traditional e-tailer first pay the wholesale price per unit of product, *w*, to the supplier, and then the consumer pays the retail price per unit of product, *p*, to both types of e-tailer at the same time as he or she purchases the product.

In this paper, the demand of the two electronic channels is determined by the market size *a*, the market share of consumers' preference for e-tail channels with blockchain (called

"blockchain-sensitive") θ , the cross-effort elasticity coefficient β and the effort of ER_1 or ER_2 (i.e., t_b or t_e). From the research of Cole R et al. [12], we know that the degree of consumer trust in the product is a key factor influencing the purchase decision. Thus, we take the platform's efforts to gain consumer trust as a factor that directly affects channel demand regardless of price [42]. The modified demand functions of the two channels are shown as follows:

$$D^{B} = \theta a + t_{b} - \beta t_{e} \tag{1}$$

$$D^{E} = (1 - \theta)a + t_{e} - \beta t_{h} \tag{2}$$

where superscript *B* or subscript *b* denotes the e-tail channel with blockchain, superscript *E* or subscript *e* denotes the e-tail channel without blockchain and *D* is the basic product demand. θ , which reflects consumers' preference for e-tail channels with blockchain. Conversely, $1-\theta$ reflects consumers' preference for e-tail channels without blockchain. β is the cross-effort elasticity coefficient between the product effort and demand for an alternative product. Meanwhile, the channel demand is directly proportional to the channel's own effort and inversely proportional to the effort of the other channel. For analytical convenience, we assume that the effort elasticity coefficient of self-effort is equal to 1, and the demand is impacted by the product effort of another channel less than it is by its channel effort (i.e., $0 < \beta < 1$).

4. Equilibria analysis

In this section, we present the equilibria for S, ER_1 and ER_2 , which are all played at Stackelberg with the supplier being the leader. Therefore, the supplier informs the chain to produce and distribute a certain product and sell at a wholesale price w. In channel B (i.e., e-tail channel with blockchain), firm ER_1 seeks to carry out the supply chain on a blockchain. Thus, firm ER_1 needs to make effort t_b . Meanwhile, in channel E (i.e., e-tail channel without blockchain), although firm ER_2 adopts a traditional online platform for supply chain negotiations rather than using blockchain technology, there are still other efforts t_e to pay. Finally, supply chain transactions occur on platforms, and the games evolve according to the following steps: The supplier will be able to deliver products at a wholesale price w; the two e-tailers react to this determine by setting the same retail price p, which is exogenous, and making optimal effort t_b and t_e .

Based on the above, the profit functions (Π_S , $\Pi^B_{ER_1}$ and $\Pi^E_{ER_2}$) of the supplier and two e-tailers are given by Eqs (3), (4) and (5), respectively:

$$\Pi_{s} = (w - \alpha t_{b})D^{B} + \gamma_{b} + wD^{E}$$
(3)

$$\Pi^{B}_{ER_{1}} = (p - w - \alpha t_{b})D^{B} + \gamma_{b}$$

$$\tag{4}$$

$$\Pi_{ER_2}^E = (p - w - \alpha t_e) D^E \tag{5}$$

It is worth noting that the supplier's profit includes two major parts: the profit obtained from the blockchain channel and the profit obtained from the traditional channel without blockchain technology. When the e-tailer implements blockchain technology in the electronic channel, the upstream supplier must also join it. Therefore, the operating cost and the brand value effect of implementing the blockchain exist in the supplier's profit function at the same time. However, in the traditional channel without blockchain technology, since channel operations are mainly implemented by e-tailers, the supplier's operating costs are not considered. To obtain the optimal decision, we substitute (1), (2) into (3), (4), and (5); the second derivative of t_b in (4) and the second derivative of t_e in (5) are less than zero. Let the first derivative of t_b and the first derivative of t_e equal zero; then, we obtain t_b^* and t_e^* . Next, substituting t_b^* and t_e^* into (3), finding the first derivative of w and letting it equal zero, we can obtain the optimal wholesale price w^* . The equilibria of the game are displayed in the following theorem:

Theorem 1 The optimal firms' strategies in the game are:

$$w^* = \frac{a(4+\beta\theta(2-\beta))\alpha + 2p(1-\beta)(3-\beta)(\beta+2)}{2(1-\beta)(5-2\beta)(\beta+2)}$$
(6)

$$t_b^* = \frac{a(-(5-4\beta)(2-\beta)\theta + 4\beta^2 - 6\beta - 2)\alpha + 2p(\beta+2)(1-\beta)}{2\alpha(\beta+2)(1-\beta)(5-2\beta)}$$
(7)

$$t_{e}^{*} = \frac{a((4\beta^{2} - 15\beta + 10)\theta - 4(3 - 2\beta))\alpha + 2p(\beta + 2)(1 - \beta)}{2\alpha(\beta + 2)(1 - \beta)(5 - 2\beta)}$$
(8)

Furthermore, substituting Eqs (6), (7) and (8) into Eqs (3), (4) and (5), the optimal profits of the supplier and two e-tailers can be obtained as follows:

$$\Pi_{s}^{*} = \frac{1}{4\alpha(\beta+2)^{2}(5-2\beta)(1-\beta)} (a^{2}((10-9\beta)(2-\beta)\theta^{2}+8\beta(3-2\beta)\theta+8\beta^{2}+4\beta+4)\alpha^{2} + (a(-4\beta(\beta+2)(1-\beta)\theta+4(2\beta+3)(\beta+2)(1-\beta))p-4\gamma_{b}(\beta+2)^{2}(2\beta-5)(1-\beta))\alpha + 4p^{2}(\beta+2)^{2}(1-\beta)^{2})$$
(9)

$$\Pi_{ER_{1}}^{B*} = \frac{1}{4\alpha(5-2\beta)^{2}(\beta+2)^{2}} \left(a^{2}(25(2-\beta)^{2}\theta^{2}+20(2\beta-1)(2-\beta)\theta+4(2\beta-1)^{2})\alpha^{2} + (2p(1-\beta)(2-\beta)(2+\beta)\theta+8p(1-\beta)(2\beta-1)(\beta+2))a+4\gamma_{b}(\beta+2)^{2}(5-2\beta)^{2})\alpha^{(10)} + 4p^{2}(\beta+2)^{2}(1-\beta)^{2}\right)$$

$$\Pi_{ER_{2}}^{E*} = \frac{\left(a(3\beta\theta - 10\theta + 8)\alpha + 2p(\beta + 2)(1 - \beta)\right)^{2}}{4\alpha(5 - 2\beta)^{2}(\beta + 2)^{2}}$$
(11)

Theorem 1 shows that both the decision variables (i.e., w^* , t_b^* and t_e^*) and the profit level of the supplier and the two e-tailers (i.e., Π_s^* , $\Pi_{ER_1}^{B*}$ and $\Pi_{ER_2}^{E*}$) are affected by parameters α , β and θ . In addition, the profit level of the supplier and the e-tailer using blockchain technology (i.e., Π_s^* and $\Pi_{ER_2}^{B*}$) are also affected by the brand value effect at the same time.

To further understand how parameters (α, β, θ) affect decision variables (i.e., w^* , t_b^* and t_e^*) and the profit level of the supplier and the two e-tailers (i.e., Π_s^* , $\Pi_{ER_1}^{B*}$ and $\Pi_{ER_2}^{E*}$), we will conduct a specific analysis of the properties of decision variables and profits.

4.1 The property analysis for the decision variables

4.1.1 The property analysis for the optimal *W***. Property 1.** With the ascension of α , β , and θ , the optimal wholesale price will increase. Specifically, three cases are possible:

(i) $\frac{\partial w^*}{\partial \alpha} > 0$; (ii) $\frac{\partial w^*}{\partial \beta} > 0$; (iii) $\frac{\partial w^*}{\partial \theta} > 0$.

Property 1 shows that the optimal wholesale prices always increase with increasing cost sensitivity to effort, the cross-effort elasticity coefficient of demand or the market share of blockchain-sensitive consumers. The reason for this result may be that when the cost sensitivity to effort increases, the costs of the electronic platform will increase more with increasing effort. To obtain more profits, e-tailers can only further increase prices. The supplier predicts that e-tailers' prices will rise, so the supplier increases wholesale prices. Demand may decrease when the cross-effort elasticity coefficient of demand increases; thus, the supplier will increase wholesale prices to maintain a higher level of profits. E-tailers tend to implement blockchain technology when the market share of blockchain-sensitive consumers increases, and the supplier further increases wholesale prices to obtain more profits in this situation.

4.1.2 The property analysis for the optimal t_b . **Property 2.** With the ascension of α and θ , the optimal effort of ER_1 will decrease. However, the influence of β on the optimal effort of ER_1 depends on the value range of α . Specifically, three cases are possible:

(i) $\frac{\partial t_b^*}{\partial \alpha} < 0$; (ii) If $0 < \alpha < -\frac{B_1}{A_1}$, therefore, $\frac{\partial t_b^*}{\partial \beta} > 0$; If $-\frac{B_1}{A_1} \le \alpha < 1$, therefore, $\frac{\partial t_b^*}{\partial \beta} \le 0$; (iii) $\frac{\partial t_b^*}{\partial \alpha} < 0$.

Property 2 shows that the effort with blockchain of ER_1 always decreases with the cost sensitivity to effort or the market share of blockchain-sensitive consumer increases. However, when the cost sensitivity to effort is small, the effort with blockchain of ER_1 always increases as the cross-effort elasticity coefficient of demand increases; when the cost sensitivity to effort is large, the effort with blockchain of ER_1 will decrease as the cross-effort elasticity coefficient of demand increases.

The reason for this result may be that when the cost sensitivity to effort increases, the cost of the electronic platform will increase more with the increase of effort, and the e-tailer will reduce effort to implement blockchain technology to lower the cost as much as possible. When the cost sensitivity to effort is small, the cost of the electronic platform will increase less with the increase of effort, so the e-tailer is willing to make greater effort in blockchain technology. Therefore, the e-tailer is still willing to make greater effort in blockchain technology even if as the cross-effort elasticity coefficient of demand increases; in contrast, when the cost sensitivity to effort. In this case, the e-tailer will reduce effort to implement blockchain technology as the cross-effort elasticity coefficient of demand increases considering the larger costs. However, when the market share of blockchain-sensitive consumers increases, the e-tailer may reduce effort to implement blockchain technology because of the larger consumer's market share.

4.1.3 The property analysis for the optimal t_e . **Property 3.** With the ascension of α , the optimal effort of ER_2 will decrease. However, the influence of β on the optimal effort of ER_2 depends on the value range of α , and the influence of θ on the optimal effort of ER_2 depends on the value range of β . Specifically, three cases are possible:

(i)
$$\frac{\partial t_e}{\partial \alpha} < 0$$

(ii) If
$$0 < \alpha < -\frac{B_2}{A_2}$$
, therefore, $\frac{\partial t_e^*}{\partial \beta} > 0$; If $-\frac{B_2}{A_2} \le \alpha < 1$, therefore, $\frac{\partial t_e^*}{\partial \beta} \le 0$;
(iii) If $0 < \beta < 0.8672$, therefore, $\frac{\partial t_e^*}{\partial \theta} > 0$; if $0.8672 \le \beta < 1$, therefore, $\frac{\partial t_e^*}{\partial \theta} \le 0$.

Property 3 shows that the effort without blockchain of ER_2 always decreases as the cost sensitivity to effort increases. When the cost sensitivity to effort is small, the effort without blockchain of ER_2 always increases as the cross-effort elasticity coefficient of demand increases; when the cost sensitivity to effort is large, the effort without blockchain of ER_2 will decrease as the cross-effort elasticity coefficient of demand increases. The above properties are the same as property 2; the only difference is that when the cross-effort elasticity coefficient of demand is large enough, the effort without blockchain of ER_2 always decreases as the market share of blockchain-sensitive consumers increases; when the cross-effort elasticity coefficient of

	α	β	θ
$w^*\uparrow$	1	1	1
$t_b^*\uparrow$	\downarrow	$\uparrow (0 < \alpha < -\frac{B_1}{A_1})$	Ļ
		$\downarrow (-\frac{B_1}{A_1} \le \alpha < 1)$	
$t_e^* \uparrow$	Ļ	$ \begin{array}{l} \uparrow (0 < \alpha < -\frac{B_2}{A_2}) \\ \downarrow (-\frac{B_2}{A_2} \le \alpha < 1) \end{array} $	$ \uparrow(0<β<0.8672) $ ↓(0.8672 ≤β<1)

Table 5. Sensitivity analysis towards the critical parameters α , β , and θ .

demand is not large enough, the effort without blockchain of ER_2 always increases as the market share of blockchain-sensitive consumers increases.

The reason for this result may be that when the cost sensitivity to effort increases, the cost of the electronic platform will increase more as the effort increases, and the traditional e-tailer will reduce effort to lower costs as much as possible. When the cost sensitivity to effort is small, the cost of the electronic platform will increase less with the increases of effort, the traditional e-tailer is willing to make greater effort to sell better products, so even if the cross-effort elasticity coefficient of demand increases, the traditional e-tailer is still willing to make greater effort to sell better products; in contrast, when the cost sensitivity to effort is greater, the cost of electronic platform will increase more with the increase of effort. Under this circumstance, the traditional e-tailer will reduce effort as the cross-effort elasticity coefficient of demand increases when considering larger costs. When the cross-effort elasticity coefficient of demand is large enough, it means that effort to implement blockchain has a greater impact on the demand of this channel. Under this circumstance, as the proportion of blockchain-sensitive consumers increases, it is difficult for the traditional e-tailer to obtain more demand even if effort increases, so the traditional e-tailer decides not to implement effort; when the crosseffort elasticity coefficient of demand is not large enough, it means that effort to implement blockchain does not have a great impact on the demand of this channel. In this case, as the proportion of blockchain-sensitive consumers increases, the traditional e-tailer decides to increase effort to obtain more demand.

To show the influence of parameters (α , β , θ) on decision variables (w^* , t_b^* and t_e^*) more clearly, we use the following table to analyse sensitivity. Table 5 summarizes the results.

4.2 The property analysis for the optimal profits

4.2.1 The property analysis for the optimal Π_s **. Property 4.** The influence of α , β , and θ on the optimal profit of the supplier depends on the value range of α . Specifically, three cases are possible:

(i) If
$$0 < \alpha < \sqrt{-\frac{B_3}{A_3}}$$
, therefore, $\frac{\partial \Pi_s^s}{\partial \alpha} < 0$; If $\sqrt{-\frac{B_3}{A_3}} \le \alpha < 1$, therefore, $\frac{\partial \Pi_s^s}{\partial \alpha} \ge 0$;
(ii) If $0 < \alpha < \frac{-B_4 + \sqrt{(B_4)^2 - 4A_4C_4}}{2A_4}$, therefore, $\frac{\partial \Pi_s^s}{\partial \beta} < 0$;
If $\frac{-B_4 + \sqrt{(B_4)^2 - 4A_4C_4}}{2A_4} \le \alpha < 1$, therefore, $\frac{\partial \Pi_s^s}{\partial \beta} \ge 0$;
(iii) If $0 < \alpha < -\frac{B_5}{A_5}$, therefore, $\frac{\partial \Pi_s^s}{\partial \theta} < 0$; if $-\frac{B_5}{A_5} \le \alpha < 1$, therefore, $\frac{\partial \Pi_s^s}{\partial \theta} \ge 0$.

Property 4 shows that when the cost sensitivity to effort is in a small range, the supplier's optimal profit level will decrease with the cost sensitivity to effort, and the cross-effort elasticity coefficient of demand and the market share of blockchain-sensitive consumers increases. However, when the cost sensitivity to effort is in a larger range, the supplier's profit level will increase as the cost sensitivity to effort, the cross-effort elasticity coefficient of demand and the

market share of blockchain-sensitive consumers increase. Property 4 further illustrates that the important factor that affects the supplier's profit level is cost sensitivity to effort.

The reason for this result may be that when the cost sensitivity to effort increases in a smaller range, the cost of the electronic platform will increase with increasing effort. Under this circumstance, the e-tailer's costs gradually rise, resulting in a decrease in the profit level, which in turn leads to a decrease in the profit level of the supplier. However, when the cost sensitivity to effort increases in a larger range, because the cost of the electronic platform is too high, the e-tailer will reduce effort instead to lower the cost of the platform effort. Therefore, the profit level of the supplier increases.

4.2.2 The property analysis for the optimal $\Pi^{B}_{ER_{1}}$ **. Property 5.** The influence of α and β on the optimal profit of ER_{1} depends on the value range of α . However, with the ascension of θ , the optimal profit of ER_{1} will increase. Specifically, three cases are possible:

(i) If
$$0 < \alpha < -\frac{B_6}{A_6}$$
, therefore, $\frac{\partial \Pi_{ER_1}^{B*}}{\partial \alpha} < 0$; If $-\frac{B_6}{A_6} \le \alpha < 1$, therefore, $\frac{\partial \Pi_{ER_1}^{B*}}{\partial \alpha} \ge 0$;
(ii) If $0 < \alpha < -\frac{B_8}{A_8}$, therefore, $\frac{\partial \Pi_{ER_1}^{B*}}{\partial \beta} < 0$; If $-\frac{B_8}{A_8} \le \alpha < 1$, therefore, $\frac{\partial \Pi_{ER_1}^{B*}}{\partial \beta} \ge 0$;
(iii) $\frac{\partial \Pi_{ER_1}^{B*}}{\partial \theta} > 0$.

Property 5 shows that when the cost sensitivity to effort is in a small range, the optimal profit level of the e-tailer implementing blockchain technology will decrease with the cost sensitivity to effort or the cross-effort elasticity coefficient of demand increases. However, when the cost sensitivity to effort or the cross-effort elasticity coefficient of demand increases. The reason for this result may be that when the cost sensitivity to effort increases in a smaller range, the e-tailer's cost gradually increases, resulting in a decrease in profit. However, when the cost sensitivity to effort is increased in a larger range, because the effort cost of the electronic platform is too high, the e-tailer will reduce the effort to lower the effort cost of the platform. Therefore, the e-tailer's profit level increases. This is similar to property 4.

Unlike property 4, regardless of the cost sensitivity to effort, the optimal profit level of an etailer implementing blockchain technology will always increase as the market share of blockchain-sensitive consumers increases. The reason for this result may be that there are more blockchain consumers in the market when the proportion of the blockchain-sensitive consumer market increases, and the e-tailer can gain more market share when implementing blockchain technology, therefore bringing the rise of profits.

4.2.3 The property analysis for the optimal $\Pi_{ER_2}^E$. **Property 6.** Similar to property 5, the influence of α and β on the optimal profit of ER_2 depends on the value range of α . In contrast to property 5, with the ascension of θ , the optimal profit of ER_2 will decrease. Specifically, three cases are possible:

(i) If
$$0 < \alpha < -\frac{B_{11}}{A_{11}}$$
, therefore, $\frac{\partial \Pi_{ER_2}^{E_R}}{\partial \alpha} < 0$; If $-\frac{B_{11}}{A_{11}} \le \alpha < 1$, therefore, $\frac{\partial \Pi_{ER_2}^{E_R}}{\partial \alpha} \ge 0$;
(ii) If $0 < \alpha < -\frac{B_{13}}{A_{13}}$, therefore, $\frac{\partial \Pi_{ER_2}^{E_R}}{\partial \beta} < 0$; If $-\frac{B_{13}}{A_{13}} \le \alpha < 1$, therefore, $\frac{\partial \Pi_{ER_2}^{E_R}}{\partial \beta} \ge 0$;
(iii) $\frac{\partial \Pi_{ER_2}^{E_R}}{\partial \theta} < 0$.

Property 6 shows that when the cost sensitivity to effort is in a small range, the optimal profit level of the traditional e-tailer will decrease with the cost sensitivity to effort or the cross-effort elasticity coefficient of demand increases. When the cost sensitivity to effort is in a larger range, the profit level of the traditional e-tailer will increase as the cost sensitivity to effort or the cross-effort elasticity coefficient of demand increase. The reason for this result may be that when the cost sensitivity to effort increases in a smaller range, the cost of the electronic

	α	β	θ	
Π^*_{S} \uparrow	$\downarrow (0 < lpha < \sqrt{-rac{B_3}{A_3}})$	$\downarrow (0 < lpha < rac{-B_4 + \sqrt{\Delta_4}}{2A_4})$	$\downarrow (0 < lpha < \sqrt{-rac{B_5}{A_5}})$	
	$\uparrow (\sqrt{-rac{B_3}{A_3}} \le lpha < 1)$	$\uparrow (rac{-B_4+\sqrt{\Delta_4}}{2A_4}\leq lpha <1)$	$\uparrow (\sqrt{-rac{B_5}{A_5}} \le lpha < 1)$	
$\overline{\Pi^{\scriptscriptstyle B*}_{\scriptscriptstyle ER_1}}$ \uparrow	$\downarrow (0 < lpha < -rac{B_{6}}{A_{6}})$	$\downarrow (0 < lpha < -rac{B_8}{A_8})$	1	
	$\uparrow (-rac{B_6}{A_6} \leq lpha < 1)$	$\uparrow (-rac{B_8}{A_8} \leq lpha < 1)$		
$\Pi^{E*}_{ER_2}$ \uparrow	$\downarrow (0 < \alpha < -\frac{B_{11}}{A_{11}})$	$\downarrow (0 < \alpha < -\frac{B_{13}}{A_{13}})$	Ļ	
	$\uparrow (-\frac{B_{11}}{A_{11}} \leq \alpha < 1)$	$\uparrow (-rac{B_{13}}{A_{13}} \leq lpha < 1)$		

Table 6. Sensitivity analysis towards the critical parameters α , β , and θ .

https://doi.org/10.1371/journal.pone.0297978.t006

platform will increase with increasing effort. For this time, the cost of the traditional e-tailer will gradually increase, resulting in a decrease in profit; when the cost sensitivity to effort increases on a larger scale, the traditional e-tailer will reduce effort to reduce the cost of effort of the platform due to the high cost of effort of the electronic platform. Therefore, the profit level of the traditional e-tailer increases. This is similar to property 5.

Different from property 5, regardless of the cost sensitivity to effort, the optimal profit level of traditional e-tailers always increases as the market share of blockchain-sensitive consumers increases. The reason for this result may be that there are more blockchain consumers in the market when the proportion of the blockchain-sensitive consumer market increases, and the traditional (not implementing blockchain technology) e-tailer can only obtain a smaller market share. This leads to a reduction in the level of profit.

To show the influence of parameters (α , β , θ) on the optimal profits (Π_{s}^{*} , $\Pi_{ER_{1}}^{B*}$ and $\Pi_{ER_{2}}^{E*}$) more clearly, we use the following table to analyse the sensitivity. Table 6 summarizes the results.

4.3 Comparison analysis

4.3.1 The comparison analysis for the optimal t_b and t_e . **Property 7.** The comparison results between the optimal effort of ER_1 and ER_2 depend on the value range of θ . Specifically, two cases are possible:

(i) If $0 < \theta < \frac{1}{2}$, therefore, $t_b^* > t_e^*$; (ii) If $\frac{1}{2} \le \theta < 1$, therefore, $t_b^* \le t_e^*$.

Property 7 shows that when the blockchain-sensitive consumer market accounts for less than half of the market share, the effort of the new e-tailer who implements blockchain technology will be greater than the effort of the traditional e-tailer who does not implement blockchain technology; in contrast, when the blockchain-sensitive consumer market accounts for more than half of the market share, the effort of the new e-tailer who implements blockchain technology will be less than the effort of the traditional e-tailer who does not implement blockchain technology. The reason for this result may be that when the blockchain-sensitive consumer market accounts for less than half of the market share, the new e-tailer implementing blockchain technology wants to work harder to gain a larger share of the consumer market. In contrast, less effort is required.

4.3.2 The comparison analysis for the optimal $\Pi_{ER_1}^B$ and $\Pi_{ER_2}^E$. **Property 8.** The comparison results between the optimal profit of ER_1 and ER_2 depend on the value range of α , θ , and γ_b . Specifically, three cases are possible:

(i) If
$$0 < \theta < \frac{1}{2}$$
 and $0 < \gamma_b < \frac{2ap(1-\beta)(1-2\theta)}{(5-2\beta)(\beta+2)}$, therefore, $\Pi_{ER_1}^{B*} < \Pi_{ER_2}^{E*}$;
(ii) If $0 < \theta < \frac{1}{2}$ and $\gamma_b \ge \frac{2ap(1-\beta)(1-2\theta)}{(5-2\beta)(\beta+2)}$:
when $0 < \alpha < -\frac{B_{16}}{A_{16}}$, therefore, $\Pi_{ER_1}^{B*} > \Pi_{ER_2}^{E*}$; when $-\frac{B_{16}}{A_{16}} \le \alpha < 1$, $\Pi_{ER_1}^{B*} \le \Pi_{ER_2}^{E*}$;

(iii) If $\frac{1}{2} \leq \theta < 1$, therefore, $\Pi_{ER_1}^{B*} \geq \Pi_{ER_2}^{E*}$.

where $A_{16} = a^2(-2\beta\theta^2 + (5\beta + 6)\theta - 2\beta - 3)$,

 $B_{16} = 2ap(\beta + 2)(1 - \beta)(2\theta - 1) + \gamma_b(5 - 2\beta)(\beta + 2)^2$. Property 8 shows that when the market share of blockchain-sensitive consumers and the brand value effect with blockchain are both in a small range, the profit level of the new e-tailer who implements blockchain technology is less than that of the traditional e-tailer who does not implement blockchain technology. When the market share of blockchain-sensitive consumers is relatively small but the brand value effect with blockchain is large, if the cost sensitivity to effort is in a small range, the profit level of the new e-tailer is greater than that of the traditional e-tailer that does not implement blockchain technology; conversely, if the cost sensitivity to effort is in a larger range, the profit level of the new e-tailer that implements blockchain technology is less than that of nonimplementation areas. When the blockchain-sensitive consumer market accounts for a relatively large market, the profit level of the new e-tailer that implements blockchain technology is always greater than that of the traditional e-tailer that blockchain technology is always greater than that of the traditional e-tailer that implement blockchain technology is always greater than that of the traditional e-tailer that does not implement blockchain technology.

5. Extended models

In the basic model, we consider a fixed-price wholesale model in which the supplier first simply sells their products to e-platforms, and then e-tailers sell their products to consumers on the platforms. However, the supplier will enter into agency selling agreements with e-platforms in many cases. Under this scenario, the supplier is required to share a percentage of the revenue as an agency fee for selling their products on the e-platform. How will such agency selling arrangements affect the e-tailers' optimal effort and profit in the context of blockchain? To address these issues, we further explore them by building the following model.

In this scenario, the supplier enters into an agency agreement with two-sided e-tailers. The supplier can sell products directly to consumers through the blockchain e-platform and the traditional e-platform and share revenue with two-sided e-tailers as platform agency fees.

Based on this scenario, the profit functions (Π_s^A , $\Pi_{ER_1}^A$ and $\Pi_{ER_2}^A$) of the supplier and two e-tailers are given by Eqs (12), (13) and (14), respectively:

$$\Pi_{s}^{A} = ((1 - \xi_{1})p_{s} - \alpha t_{b}^{A})D^{B} + \gamma_{b} + (1 - \xi_{2})p_{s}D^{E}$$
(12)

$$\Pi^{A}_{ER_{1}} = (\xi_{1}p_{S} - \alpha t^{A}_{b})D^{B} + \gamma_{b}$$
(13)

$$\Pi^{A}_{ER_{2}} = (\xi_{2}p_{s} - \alpha t^{A}_{e})D^{E}$$
(14)

where $0 < \xi_1 < 1$ and $0 < \xi_2 < 1$ represent the percentage of revenue shared by the supplier to the new e-tailer and traditional e-tailer, respectively. Not surprisingly, $0 < 1 - \xi_1 < 1$ and $0 < 1 - \xi_2 < 1$ represent the remaining revenues that the supplier keeps after sharing the needed proportions. Therefore, we have Theorem 2.

Theorem 2 The optimal firms' strategies in the game are:

$$t_{b}^{A*} = \frac{-a((2-\beta)\theta + \beta)\alpha + (2\xi_{1} + \beta\xi_{2})p_{s}}{\alpha(\beta+2)(2-\beta)}$$
(15)

$$t_{e}^{A*} = \frac{-a(\beta\theta + 2(1-\theta))\alpha + (\beta\xi_{1} + 2\xi_{2})p_{s}}{\alpha(\beta+2)(2-\beta)}$$
(16)

Furthermore, substituting Eqs (15) and (16) into Eqs (12), (13) and (14), the optimal profits of

the supplier and two e-tailers can be obtained as follows:

$$\Pi_{S}^{A*} = \frac{A_{17}\alpha^{2} + B_{17}\alpha + C_{17}}{\alpha(\beta+2)^{2}(2-\beta)^{2}}$$
(17)

$$\Pi_{ER_1}^{A*} = \frac{A_{18}\alpha^2 + B_{18}\alpha + C_{18}}{\alpha(\beta+2)^2(2-\beta)^2}$$
(18)

$$\Pi_{ER_2}^{A*} = \frac{\left(a(\beta\theta + 2(1-\theta)\alpha - p_s(\beta^2\xi_2 + \beta\xi_1 - 2\xi_2)\right)}{\alpha(\beta + 2)^2(2-\beta)^2}$$
(19)

where $A_{17} = a^2(\beta\theta - \beta - 2\theta)^2$,

$$B_{17} = \beta^4 \gamma_b - ap_s(1 - \theta\xi_2)\beta^3 - 2(ap_s + 4\gamma_b)\beta^2 + 4ap_s(1 - 2\theta\xi_2 - (1 - \theta)\xi_1)\beta - 8a((1 - \theta)\xi_2 + \theta\xi_1 - 1)p_s + 16\gamma_b)$$

$$\begin{split} C_{17} &= -p_s^2((\xi_1^2 + \xi_2^2 - \xi_1 - \xi_2)\beta^4 + (\xi_1\xi_2 - \xi_1 - \xi_2)\beta^3 + (-8\xi_1^2 - 7\xi_2^2 + 6\xi_1 + 6\xi_2)\beta^2 \\ &+ (-8\xi_1\xi_2 + 4\xi_1 + 4\xi_2)\beta + 12\xi_1^2 + 8\xi_2^2 - 8\xi_1 - 8\xi_2) \end{split};$$

$$A_{18} - A_{17}$$

$$\begin{split} B_{18} &= \gamma_b \beta^4 - 2a p_s \xi_1 (1-\theta) \beta^3 - (2a p_s (2\theta \xi_1 + (1-\theta) \xi_2) + 8\gamma_b) \beta^2 + 4a p_s ((1-\theta) \xi_1 - \theta \xi_2) \beta + 8(a \theta \xi_1 p_s + 2\gamma_b) \end{split}$$

$$C_{18} = p_s^2 (\beta^2 \xi_1 + \beta \xi_2 - 2\xi_1)^2.$$

Through the property of the previous basic model, we find that whether it is the supplier or the two e-tailers, the key influence on the profitability of the supply chain members is ultimately the cost sensitivity to effort α . Therefore, to simplify our study and strengthen the core results of this paper, we will further conduct a specific analysis of how parameters α affect decision variables (i.e., T_b^{A*} and t_e^{A*}) and the profit level of the supply chain members (i.e., Π_s^{A*} , $\Pi_{ER_1}^{A*}$ and $\Pi_{ER_2}^{A*}$).

5.1 The property analysis for the decision variables

5.1.1 The property analysis for the optimal t_b^A and t_e^A . **Property 9.** With the ascension of α , the optimal effort of ER_1 and ER_2 will decrease. Specifically, two cases are possible:

(i) $\frac{\partial t_b^{A^*}}{\partial \alpha} < 0$; (ii) $\frac{\partial t_c^{A^*}}{\partial \alpha} < 0$. Similar to properties 2 and 3, property 9 still shows our core results. This implies that under the agency scenario, the effort with blockchain of ER_1 always decreases with the cost sensitivity to effort or the market share of blockchain-sensitive consumers increases. Not surprisingly, the effort without blockchain of ER_2 is negatively correlated with the cost sensitivity to effort.

Corollary 1. (i) $\frac{\partial t_b^{A^*}}{\partial \alpha} < \frac{\partial t_b^*}{\partial \alpha}$; (ii) $\frac{\partial t_c^{A^*}}{\partial \alpha} < \frac{\partial t_c^*}{\partial \alpha}$.

It is worth noting that Corollary 1 is also a very strong result. By comparing the agency scenario with the basic scenario, we find that cost sensitivity to effort has a greater impact on the e-tailer's effort in the agency scenario. This is an interesting finding and reveals that e-tailers in the agency scenario may be more affected by costs due to the level of profit they receive.

5.2 The property analysis for the optimal profits

5.2.1 The property analysis for the optimal Π_s^A **. Property 10.** The influence of α on the optimal profit of the supplier depends on the value range of α . Specifically, two cases are possible:

(i) If
$$0 < \alpha < \sqrt{-\frac{B_{19}}{A_{19}}}$$
, therefore, $\frac{\partial \Pi_S^{A*}}{\partial \alpha} < 0$; (ii) If $\sqrt{-\frac{B_{19}}{A_{19}}} \le \alpha < 1$, therefore, $\frac{\partial \Pi_S^{A*}}{\partial \alpha} \ge 0$

Similar to property 4, property 10 still shows our core results. This implies that under the agency scenario, when the cost sensitivity to effort is in a smaller range, the supplier's optimal profit level will decrease with the cost sensitivity to effort. Conversely, when the cost sensitivity to effort is in a larger range, the supplier's optimal profit is positively correlated with the cost sensitivity to effort.

Corollary 2.

(i) If $0 < \alpha < \min(\sqrt{-\frac{B_3}{A_3}}, \sqrt{-\frac{B_{19}}{A_{19}}})$, therefore, $\frac{\partial \Pi_S^*}{\partial \alpha} > \frac{\partial \Pi_S^*}{\partial \alpha}$; (ii) If $\max(\sqrt{-\frac{B_3}{A_3}}, \sqrt{-\frac{B_{10}}{A_{19}}}) \le \alpha < 1$, therefore, $\frac{\partial \Pi_S^*}{\partial \alpha} \le \frac{\partial \Pi_S^*}{\partial \alpha}$. Corollary 2 is also a very strong result. By comparing the agency scenario with the basic scenario, we find that when the cost sensitivity to effort is in a smaller range, the profit of the supplier in the agency scenario decreases less than the profit of the supplier in the basic scenario for every unit increase in the cost sensitivity to effort. Namely, cost sensitivity to effort has less impact on the supplier's profit in the agency scenario. However, when the cost sensitivity to effort is in a larger range, the profit of the supplier in the agency scenario. However, when the cost sensitivity to effort. In other words, cost sensitivity to effort also has less impact on the supplier in the basic scenario. Consequently, Corollary 2 reveals that when the cost sensitivity to effort is in a smaller range, the supplier is more profitable in the agency scenario. In contrast, when the cost sensitivity to effort is in a smaller range, the basic scenario is more beneficial for the supplier.

5.2.2 The property analysis for the optimal $\Pi_{ER_1}^A$. **Property 11.** The influence of α on the optimal profit of ER_1 depends on the value range of α . Specifically, four cases are possible:

$$\begin{split} &\text{If } 0 < \xi_1 < \frac{\beta}{2-\beta^2}, 0 < \xi_2 < \frac{\xi_1(2-\beta^2)}{\beta}; \\ &\text{(i) When } 0 < \alpha < -\frac{B_{20}}{A_{20}}, \text{therefore, } \frac{\partial \Pi_{ER_1}^{A*}}{\partial \alpha} < 0; \text{(ii) When } -\frac{B_{20}}{A_{20}} \leq \alpha < 1, \text{ therefore, } \frac{\partial \Pi_{ER_1}^{A*}}{\partial \alpha} \geq 0. \\ &\text{If } 0 < \xi_1 < \frac{\beta}{2-\beta^2}, \frac{\xi_1(2-\beta^2)}{\beta} \leq \xi_2 < 1; \end{split}$$

(iii) When $0 < \alpha < -\frac{B_{21}}{A_{21}}$, therefore, $\frac{\partial \prod_{E_{R_1}}^{A_*}}{\partial \alpha} < 0$; (iv) When $-\frac{B_{21}}{A_{21}} \leq \alpha < 1$, therefore,

 $\frac{\partial \Pi_{RR}^{A_1}}{\partial \alpha} \ge 0.$ Property 11 implies that under the agency scenario, when the cost sensitivity to effort is in a smaller range, the optimal profit level of the new e-tailer will decrease with the cost sensitivity to effort. In contrast, if the cost sensitivity to effort is in a larger range, the new e-tailer's optimal profit is positively correlated with the cost sensitivity to effort.

Corollary 3. $\frac{\partial \Pi_{ER_1}^{A*}}{\partial \alpha} > \frac{\partial \Pi_{ER_1}^{B*}}{\partial \alpha}$.

Similar to Corollary 2, by comparing the agency scenario with the basic scenario, we find that when the cost sensitivity to effort is in a smaller range, the profit of the new e-tailer in the agency scenario decreases less than the profit of the new e-tailer in the basic scenario for every unit increase in the cost sensitivity to effort. Namely, cost sensitivity to effort has less impact on the new e-tailer's profit in the agency scenario. Conversely, when the cost sensitivity to effort is in a larger range, the profit of the new e-tailer in the agency scenario increases more than the profit of the new e-tailer in the basic scenario for every unit increase in the cost sensitivity to effort. In other words, cost sensitivity to effort has a greater impact on the new e-

tailer's profit in the agency scenario. As such, Corollary 3 reveals that regardless of the scope of the cost sensitivity to effort, the new e-tailer is always more profitable in the agency scenario.

5.2.3 The property analysis for the optimal $\Pi_{ER_2}^A$. Property 12. Similar to property 11, the influence of α on the optimal profit of ER_2 depends on the value range of α . Specifically, four cases are possible:

$$\begin{split} &\text{If } 0 < \xi_2 < \frac{\beta}{2-\beta^2}, 0 < \xi_1 < \frac{\xi_2(2-\beta^2)}{\beta}; \\ &\text{(i) When } 0 < \alpha < -\frac{B_{22}}{A_{22}}, \text{therefore, } \frac{\partial \Pi_{ER_2}^{A*}}{\partial \alpha} < 0; \text{(ii) When } -\frac{B_{22}}{A_{22}} \leq \alpha < 1, \text{ therefore, } \frac{\partial \Pi_{ER_2}^{A*}}{\partial \alpha} \geq 0. \\ &\text{If } 0 < \xi_2 < \frac{\beta}{2-\beta^2}, \frac{\xi_2(2-\beta^2)}{\beta} \leq \xi_1 < 1; \\ &\text{(iii) When } 0 < \alpha < -\frac{B_{23}}{A_{23}}, \text{ therefore, } \frac{\partial \Pi_{ER_2}^{A*}}{\partial \alpha} < 0; \text{(iv) When } -\frac{B_{23}}{A_{23}} \leq \alpha < 1, \text{ therefore, } \end{split}$$

 $\frac{\partial \Pi_{ER_2}^{A*}}{\partial \alpha} \ge 0$. Property 12 implies that under the agency scenario, when the cost sensitivity to effort is in a smaller range, the optimal profit level of the traditional e-tailer will decrease with the cost sensitivity to effort. In contrast, if the cost sensitivity to effort is in a larger range, the traditional e-tailer's optimal profit is positively correlated with the cost sensitivity to effort.

Corollary 4.
$$\frac{\partial \Pi_{ER_2}^{A*}}{\partial \alpha} > \frac{\partial \Pi_{ER_2}^{E*}}{\partial \alpha}$$
.

Similar to Corollary 3, by comparing the agency scenario with the basic scenario, we find that when the cost sensitivity to effort is in a smaller range, the profit of the traditional e-tailer in the agency scenario decreases less than the profit of the traditional e-tailer in the basic scenario for every unit increase in the cost sensitivity to effort. Namely, cost sensitivity to effort has less impact on the traditional e-tailer's profit in the agency scenario. Conversely, when the cost sensitivity to effort is in a larger range, the profit of the traditional e-tailer in the agency scenario increases more than the profit of the traditional e-tailer in the basic scenario for every unit increase in the cost sensitivity to effort. In other words, cost sensitivity to effort has a greater impact on the traditional e-tailer's profit in the agency scenario. As such, Corollary 4 reveals that regardless of the scope of the cost sensitivity to effort, the traditional e-tailer is always more profitable in the agency scenario.

As we can see from properties and corollaries in the extended model, the agency scenario and the basic scenario are very similar. Consequently, the other managerial findings and insights that arise under the basic model will continue to hold under the agency model. In addition, the most important we find is that the e-tailers are always more profitable in the agency model regardless of the scope of the cost sensitivity to effort. Namely, compared to the basic model, the agency model may be a more efficient way to improve the benefits of e-tailers. However, as fa as the supplier is concerned, the agency scenario is not always favourable. Specifically, when the cost sensitivity to effort is in a smaller range, the supplier is more profitable in the agency scenario. In contrast, when the cost sensitivity to effort is in a larger range, the basic scenario is more beneficial for the supplier.

6. Numerical examples and analysis

To show the property and comparison results more clearly, we provide some numerical examples to confirm the results derived in the previous sections and display the performance of the proposed blockchain technology by comparison with two e-tail channel models. To ensure that the parameters are set appropriately and all the conditions assumed in the previous section are satisfied, we refer to the empirical estimations reported in the literature (e.g., Cole et al. [12], Sunny et al. [19], van Hoek [58]) as well as the parameter estimations from the literature of game theory (e.g., Liu et al. [24], Liu et al. [25], Genc and De Giovanni [59]). Specifically, Genc and De Giovanni [59] perform an numerical examples of the game results by setting the

relevant parameters, clearly presenting the propositional results and verifying the accuracy of the conclusions. As such, we suppose that the values of the parameters are as follows: a = 100, p = 5. As stated in Section 3, the settings of these parameters need to satisfy the following conditions: (i) $0 < \theta < 1$; (ii) $0 < \alpha < 1$; (iii) $0 < \beta < 1$; and (iv) $\gamma_b \ge 0$.

6.1 Numerical analysis of the decision variables

This section mainly examines the property in Section 5. To further clearly present how parameters (α , β , θ) affect decision variables (i.e., w^* , t_b^* and t_e^*) and the profit level of the supplier and the two e-tailers (i.e., Π_s^* , $\Pi_{ER_1}^{B*}$ and $\Pi_{ER_2}^{E*}$), we will perform a specific numerical analysis for the properties of decision variables and profits.

6.1.1 Numerical analysis of the optimal *W*. To clearly present the impact of the cost sensitivity to effort (α), cross-effort elasticity coefficient of demand (β) and the market share of blockchain-sensitive consumer (θ) on the optimal wholesale price (*w*), the relationship between $\frac{\partial w^*}{\partial \alpha}$ and β/θ , relationship between $\frac{\partial w^*}{\partial \alpha}$ and β/θ , and relationship between $\frac{\partial w^*}{\partial \theta}$ and α/β will be presented by fixing a = 100, p = 5. As shown in Figs 2–4.

The conclusion in Property 1 is demonstrated in Fig 2, showing the relationship between $\frac{\partial w^*}{\partial \alpha}$ and β/θ , and we see that the result is always positive. Specifically, setting a = 100, regardless of how the cross-effort elasticity coefficient of demand β ($0 < \beta < 1$) or the market share of blockchain-sensitive consumer θ ($0 < \theta < 1$) changes, the improvement of the cost sensitivity to effort α ($0 < \alpha < 1$) always increases the optimal wholesale price. Fig 3 shows the relationship between $\frac{\partial w^*}{\partial \beta}$ and α/β , and we can also easily find that the result is always positive. Specifically,



https://doi.org/10.1371/journal.pone.0297978.g002



Fig 3. Relationship between $\frac{\partial w^*}{\partial \beta}$ and α/β . (*a* = 100) (*a* = 100, *p* = 5, θ = 0.4). https://doi.org/10.1371/journal.pone.0297978.g003

setting a = 100, p = 5, $\theta = 0.4$, regardless of how the cost sensitivity to effort α or the cross-effort elasticity coefficient of demand β changes, the optimal wholesale price always increases with the improvement of the cross-effort elasticity coefficient of demand β . Similarly, Fig 4 shows that the optimal wholesale price grows as the market share of blockchain-sensitive consumer θ increases, regardless of whether the cost sensitivity to effort α or cross-effort elasticity coefficient of demand β changes.

6.1.2 Numerical analysis of the optimal t_b . To clearly present the impact of the cost sensitivity to effort (α), cross-effort elasticity coefficient of demand (β) and the market share of blockchainsensitive consumer (θ) on the optimal effort with blockchain of ER_1 (t_b), the relationship between $\frac{\partial t_b^*}{\partial \alpha}$ and β , relationship between $\frac{\partial t_b^*}{\partial \beta}$ and α , and relationship between $\frac{\partial t_b^*}{\partial \theta}$ and β will be presented by fixing a = 100, p = 5. As shown in Figs 5–7. The conclusion in Property 2 is demonstrated in Fig 5, showing the relationship between $\frac{\partial t_b^*}{\partial \alpha}$ and β ; we see that the result is always negative. Specifically, setting $a = 100, p = 5, \alpha = 0.4$, no matter how the cross-effort elasticity coefficient of demand β ($0 < \beta < 1$) changes, the improvement of the cost sensitivity to effort α ($0 < \alpha < 1$) does not increase the effort with a blockchain of ER_1 . Fig 6 shows the relationship between $\frac{\partial t_b^*}{\partial \beta}$ and α . Specifically, setting $a = 100, p = 5, \beta = 0.4, \theta = 0.4, when the cost sensitivity to the effort <math>\alpha$ is lower than a threshold ($\alpha = 0.0075$), the improvement of the cost sensitivity to effort α exceeds the threshold, the effort with a blockchain of ER_1 . Once the cost sensitivity to effort α exceeds the threshold, the effort with blockchain of ER_1 will not grow as cross-effort elasticity coefficient of demand β increases. Fig 7 shows that the effort with blockchain of ER_1 decreases as the market share of blockchain-sensitive consumer θ ($0 < \theta < 1$) increases, regardless of whether the cross-effort elasticity coefficient of demand β increases. Fig 7 shows that



https://doi.org/10.1371/journal.pone.0297978.g004

6.1.3 Numerical analysis of the optimal t_e . To clearly present the impact of the cost sensitivity to effort (α), cross-effort elasticity coefficient of demand (β) and the market share of blockchain-sensitive consumer (θ) on the optimal effort without blockchain of $ER_2(t_e)$, the relationship between $\frac{\partial t_{\alpha}^{*}}{\partial \alpha}$ and β , relationship between $\frac{\partial t_{\alpha}^{*}}{\partial \beta}$ and α , and relationship between $\frac{\partial t_{\alpha}^{*}}{\partial \theta}$ and β will be presented by fixing a = 100, p = 5. As shown in Figs $\frac{8}{10}$. We can easily see that $\frac{\partial t_e^*}{\partial x} < 0$ in Fig. 8. Specifically, in Fig 8, setting a = 100, p = 5, $\alpha = 0.4$, no matter how the cross-effort elasticity coefficient of demand β (0< β <1) changes, effort without blockchain of ER₂ always decreases when the cost sensitivity to the effort α (0< α <1) increases. In Fig 9, setting $a = 100, p = 5, \beta =$ 0.4, $\theta = 0.4$, when α (0< α <0.0079) is low enough, effort without blockchain of ER₂ increases with the increases of the cost sensitivity to the effort β ; however, when α (0.0079 $\leq \alpha < 1$) is not low enough, the improvement of the cross-effort elasticity coefficient of demand β (0< β <1) does not increase the effort without blockchain of ER_2 . In Fig 10, setting a = 100, when β $(0.8672 \le \beta \le 1)$ is high enough, the improvement of the market share of blockchain-sensitive consumer θ (0< θ <1) always decreases the effort without blockchain of *ER*₂; in contrast, when β $(0 < \beta < 0.8672)$ is not high enough, the improvement of the market share of blockchain-sensitive consumer θ (0< θ <1) always increases the effort without blockchain of *ER*₂.

6.2 Numerical analysis of the optimal profits

6.2.1 Numerical analysis of the optimal Π_s . To clearly present the impact of the cost sensitivity to effort (α), cross-effort elasticity coefficient of demand (β) and the market share of



blockchain-sensitive consumer (θ) on the optimal profits of supplier (Π_S), the relationship between $\frac{\partial \Pi_S^*}{\partial \alpha}$ and α , relationship between $\frac{\partial \Pi_S^*}{\partial \beta}$ and α , and relationship between $\frac{\partial \Pi_S^*}{\partial \theta}$ and α will be presented by fixing a = 100, p = 5, $\beta = 0.4$, $\theta = 0.4$. As shown in Figs 11–13.

Setting a = 100, p = 5, $\beta = 0.4$, $\theta = 0.4$ by comparing the three pictures, it is easy to find that when α is low enough, $\frac{\partial \Pi_s^*}{\partial \alpha} < 0$, $\frac{\partial \Pi_s^*}{\partial \theta} < 0$; when α is not low enough, $\frac{\partial \Pi_s^*}{\partial \alpha} \ge 0$, $\frac{\partial \Pi_s^*}{\partial \theta} \ge 0$, $\frac{\partial \Pi_s^*}{\partial \theta} \ge 0$. Specifically, in Fig 11, when α ($0 < \alpha < 0.0428$) is low enough, the supplier's profits decrease with increasing cost sensitivity to effort α ; when α ($0.0428 \le \alpha < 1$) is not low enough, the supplier's profits increase with increasing cost sensitivity to effort α . In Fig 12, when the cost sensitivity to effort α is lower than a threshold ($\alpha = 0.0222$), the improvement of the crosseffort elasticity coefficient of demand β ($0 < \beta < 1$) decreases the supplier's profits. Once the cost sensitivity to effort α exceeds the threshold, the supplier's profits will grow as the cross-effort elasticity coefficient of demand β increases. Similarly, in Fig 13, when the cost sensitivity to effort α is lower than a threshold ($\alpha = 0.0076$), the improvement of the market share of blockchain-sensitive consumer θ ($0 < \theta < 1$) decreases the supplier's profits. Once the cost sensitivity to effort α exceeds the threshold, the supplier's profits. Once the cost sensitivity to effort α exceeds the threshold ($\alpha = 0.0076$), the improvement of the market share of blockchain-sensitive consumer θ ($0 < \theta < 1$) decreases the supplier's profits. Once the cost sensitivity to effort α exceeds the threshold, the supplier's profits will grow as the market share of blockchain-sensitive consumer θ increases.

6.2.2 Numerical analysis of the optimal $\Pi_{ER_1}^{B}$. To clearly present the impact of the cost sensitivity to effort (α), cross-effort elasticity coefficient of demand (β) and the market share of blockchain-sensitive consumer (θ) on the optimal profits of ER_1 ($\Pi_{ER_1}^{B}$), the relationship between $\frac{\partial \Pi_{ER_1}^{B*}}{\partial \alpha}$ and α , relationship between $\frac{\partial \Pi_{ER_1}^{B*}}{\partial \theta}$ and α , and relationship between $\frac{\partial \Pi_{ER_1}^{B*}}{\partial \theta}$ and α/θ will be presented by fixing a = 100, p = 5, $\beta = 0.4$. As shown in Figs 14–16.















https://doi.org/10.1371/journal.pone.0297978.g014

Setting a = 100, p = 5, $\beta = 0.4$, $\theta = 0.4$, by comparing Figs 14 and 15, it is easy to find that when α is low enough, $\frac{\partial \Pi_{ER_1}^{B*}}{\partial \alpha} < 0$, $\frac{\partial \Pi_{ER_1}^{B*}}{\partial \beta} < 0$; when α is not low enough, $\frac{\partial \Pi_{ER_1}^{B*}}{\partial \alpha} \ge 0$, $\frac{\partial \Pi_{ER_1}^{B*}}{\partial \beta} \ge 0$. Specifically, in Fig 14, when the cost sensitivity to effort α is lower than a threshold ($\alpha = 0.0514$), the improvement of the cost sensitivity to effort α decreases the profits of retailer ER_1 . Once the cost sensitivity to effort α increases. Similarly, in Fig 15, when the cost sensitivity to effort α is lower than a threshold ($\alpha = 0.0791$), the improvement of the cross-effort elasticity coefficient of demand β ($0 < \beta < 1$) decreases the profits of retailer ER_1 . Once the cost sensitivity to effort α as the profits of retailer ER_1 will grow as the cost sensitivity to effort α increases. Similarly, in Fig 15, when the cost sensitivity to effort α is lower than a threshold ($\alpha = 0.0791$), the improvement of the cross-effort elasticity coefficient of demand β ($0 < \beta < 1$) decreases the profits of retailer ER_1 . Once the cost sensitivity to effort α exceeds the threshold, the profits of retailer ER_1 will grow as the cross-effort elasticity coefficient of demand β increases. In Fig 16, setting a = 100, p = 5, $\beta = 0.4$, we know that regardless of how α ($0 < \alpha < 1$) or θ ($0 < \theta < 1$) changes, the profits of retailer ER_1 always grow as the market share of blockchain-sensitive consumer θ increases.

6.2.3 Numerical analysis of the optimal $\Pi_{ER_2}^E$. To clearly present the impact of the cost sensitivity to effort (α), cross-effort elasticity coefficient of demand (β) and the market share of blockchain-sensitive consumer (θ) on the optimal profits of ER_2 ($\Pi_{ER_2}^E$), the relationship between $\frac{\partial \Pi_{ER_2}^{E*}}{\partial \alpha}$ and α , relationship between $\frac{\partial \Pi_{ER_2}^{E*}}{\partial \beta}$ and α , and relationship between $\frac{\partial \Pi_{ER_2}^{E*}}{\partial \theta}$ and α/θ will be presented by fixing a = 100, p = 5, $\beta = 0.4$. As shown in Figs 17–19.

Setting a = 100, p = 5, $\beta = 0.4$, $\theta = 0.4$, by comparing Figs <u>17</u> and <u>18</u>, it is easy to find that when α is low enough, $\frac{\partial \Pi_{ER_2}^{E*}}{\partial \alpha} < 0$, $\frac{\partial \Pi_{ER_2}^{E*}}{\partial \beta} < 0$; when α is not low enough, $\frac{\partial \Pi_{ER_2}^{E*}}{\partial \alpha} \ge 0$, $\frac{\partial \Pi_{ER_2}^{E*}}{\partial \beta} \ge 0$. Specifically, in Fig <u>17</u>, when the cost sensitivity to effort α is lower than a threshold ($\alpha = 0.0321$),





https://doi.org/10.1371/journal.pone.0297978.g016



the improvement of the cost sensitivity to effort α decreases the profits of retailer ER_2 . Once the cost sensitivity to effort α exceeds the threshold, the profits of retailer ER_2 will grow as the cost sensitivity to effort α increases. Similarly, in Fig 18, when the cost sensitivity to effort α is lower than a threshold ($\alpha = 0.1169$), the improvement of the cross-effort elasticity coefficient of demand β ($0 < \beta < 1$) decreases the profits of retailer ER_2 Once the cost sensitivity to effort α exceeds the threshold, the profits of retailer ER_2 will grow as the cross-effort elasticity coefficient of demand β increases. In Fig 19, setting a = 100, p = 5, $\beta = 0.4$, we know that regardless of how α ($0 < \alpha < 1$) or θ ($0 < \theta < 1$) changes, the profits of retailer ER_2 will not grow as the market share of blockchain-sensitive consumer θ increases.

6.3 Numerical analysis of comparison

This section mainly examines the comparison results between the optimal benefit of the *B* model and the *E* model due to blockchain technology and the situation without blockchain. We provide various numerical examples to demonstrate the reliability of our model.

6.3.1 Numerical analysis between t_b^* and t_e^* . To clearly present the impact of the market share of blockchain-sensitive consumer (θ) on $t_b^* - t_e^*$, the relationship between $t_b^* - t_e^*$ and θ will be presented by fixing a = 100, $\beta = 0.4$. As shown in Fig 20.

In Fig 20, setting a = 100, $\beta = 0.4$, we find that when the market share of blockchain-sensitive consumer θ ($0 < \theta < 1$) is comparatively low, effort with blockchain of ER_1 is greater than effort without blockchain of ER_2 ; in contrast, when the market share of blockchain-sensitive consumer θ ($0 < \theta < 1$) is comparatively high, effort without blockchain of ER_2 is greater than effort with blockchain of ER_1 .



https://doi.org/10.1371/journal.pone.0297978.g019



https://doi.org/10.1371/journal.pone.0297978.g020

6.3.2 Numerical analysis between $\Pi_{ER_1}^{B*}$ and $\Pi_{ER_2}^{E*}$. To clearly present the impact of the cost sensitivity to effort (α), cross-effort elasticity coefficient of demand (β) and the market share of blockchain-sensitive consumer (θ) on $\Pi_{ER_1}^{B*} - \Pi_{ER_2}^{E*}$, the relationship between $\Pi_{ER_1}^{B*} - \Pi_{ER_2}^{E*}$ and α , will be presented by fixing a = 100, p = 5, $\beta = 0.4$, $\theta = 0.4/0.6$, $\gamma_b = 10/20$. As shown in Figs 21–23.

Setting a = 100, p = 5, $\beta = 0.4$, $\theta = 0.4$, by comparing Figs 21 and 22, we find that when the market share of blockchain-sensitive consumer θ ($\theta = 0.4$) and the brand value effect with blockchain γ_b ($\gamma_b = 10$) are comparatively low, no matter how α changes, the profits of ER_1 are always less than the profits of ER_2 . However, when the market share of blockchain-sensitive consumer θ ($\theta = 0.4$) is comparatively low, but the brand value effect with blockchain γ_b ($\gamma_b = 20$) is comparatively high, when α ($0 < \alpha < 0.0269$) is low enough, the profits of ER_1 are greater than the profits of ER_2 ; when α ($0.0269 \le \alpha < 1$) are not low enough, the profits of ER_1 are less than the profits of ER_2 . In Fig 23, when the market share of blockchain-sensitive consumer θ ($\theta = 0.6$) is comparatively high, regardless of how the brand value effect with blockchain γ_b ($\gamma_b > 0$) or the cost sensitivity to effort α ($0 < \alpha < 1$) changes, the profits of ER_1 are always greater than the profits of ER_2 .

6.4 Numerical analysis of extended model

First, we suppose in this section that the values of the parameters are as follows: a = 100, p = 5, $p_s = 4$, $\beta = 0.4$, $\theta = 0.4$, $\xi_1 = 0.3$, and $\xi_2 = 0.3$. Then, we verify the effect of α ($0 < \alpha < 1$) on $\Delta \frac{\partial \Pi_s}{\partial \alpha}$, $\Delta \frac{\partial \Pi_{ER_1}}{\partial \alpha}$, and $\Delta \frac{\partial \Pi_{ER_2}}{\partial \alpha} (\Delta \frac{\partial \Pi_s}{\partial \alpha} = \frac{\partial \Pi_s^{**}}{\partial \alpha}, \Delta \frac{\partial \Pi_{ER_1}}{\partial \alpha} = \frac{\partial \Pi_{ER_1}^{**}}{\partial \alpha} - \frac{\partial \Pi_{ER_1}^{**}}{\partial \alpha}$, and $\Delta \frac{\partial \Pi_{ER_2}}{\partial \alpha} = \frac{\partial \Pi_{ER_2}^{**}}{\partial \alpha} - \frac{\partial \Pi_{ER_2}^{**}}{\partial \alpha}$.



From Fig 24, we can easily find that when α ($0 < \alpha < 0.0338$) is low enough, the value of $\Delta \frac{\partial \Pi_S}{\partial \alpha}$ is always positive; conversely, when α ($0.0338 \le \alpha < 1$) is not low enough, the value of $\Delta \frac{\partial \Pi_S}{\partial \alpha}$ is negative. However, in Figs 25 and 26, we know that regardless of how α ($0 < \alpha < 1$) changes, the values of $\Delta \frac{\partial \Pi_{ER_1}}{\partial \alpha}$ and $\Delta \frac{\partial \Pi_{ER_2}}{\partial \alpha}$ are always positive.

7. Discussion

7.1 Findings

In the context of the rapid development and application of blockchain, a new generation of information technology, this paper considers two competing e-tailers—a new type of e-tailer that implements blockchain technology and a traditional e-tailer that does not implement blockchain technology. We develop demand functions based on the heterogeneity of consumers' sensitivity to blockchain technology under the two channels. Using the master-slave game model to obtain equilibrium solutions, we compare the optimal effort and the optimal profit of two e-tailers. For the basic properties, interestingly, we find that there exists a critical threshold on the cost sensitivity to effort that helps each e-tailer decide whether to implement effort. If the cost sensitivity to effort is high, two-sided e-tailers will reduce their effort as much as possible to obtain greater profits. Conversely, if the cost sensitivity to effort is low, they will increase their effort to obtain more benefits.

We also discuss the role of blockchain technology in competition between e-tailers and analyse the impact of the product brand effect brought by the traceability characteristic of blockchain on the competition between e-tailers. Specifically, (i) in the comparison of optimal



https://doi.org/10.1371/journal.pone.0297978.g022

effort, the results show that when the blockchain-sensitive consumer market accounted for less than half of the market share, the effort of new e-tailers who implemented blockchain technology would be greater than the effort of traditional e-tailers who did not implement blockchain technology; in contrast, it is less. (ii) In the comparison of optimal profit, the results show that when the market share of blockchain-sensitive consumers and the brand value effect with blockchain are both in a relatively small range or when the market share of blockchain-sensitive consumers is relatively small but the brand value effect with blockchain and the cost sensitivity to the effort are large, the new e-tailer's profit is less than the traditional e-tailer's profit. Under this circumstance, the implementation of blockchain technology is not a dominant choice for the new e-tailer. When the blockchain-sensitive consumer market has a relatively small market share but the brand value effect with blockchain is large, if the cost sensitivity to the effort is in a small range, the new e-tailer obtains more benefits. In other words, the implementation of blockchain technology is beneficial for the new e-tailer. However, when the blockchain-sensitive consumer market accounts for a relatively large market, no matter how the brand value effect with blockchain and the cost sensitivity to the effort are valued within their range, the profit level of the new e-tailer is always greater than that of the traditional e-tailer. That is, the implementation of blockchain technology is always a profitable choice for the new e-tailer.

In addition, in the robustness test of the extended model, we are happy to find that all core results are still valid, which gives us greater confidence to ensure the scientific accuracy of this paper. Finally, the most important finding is that e-tailers are always more profitable in the agency model regardless of the scope of the cost sensitivity to effort. In particular, when the cost sensitivity to effort is low enough, the agency model may be a more efficient way to



https://doi.org/10.1371/journal.pone.0297978.g023

improve the benefits of supply chain members. To present the findings of this paper more clearly, we summarise the findings in Table 7.

7.2 Theoretical contributions

This paper analyses the relevant properties and finds that the cost sensitivity to effort profoundly affects e-tailers' effort implementation. Therefore, e-tailers should consider the extent of the cost sensitivity to effort when making effort implementation decisions. In particular, the if the cost sensitivity to effort is low, e-tailers should maximize their efforts to make more profits. Moreover, we find that whether e-tailers use blockchain technology depends on the brand spillover effect as well as the cost effect. In particular, e-tailers should implement blockchain technology when the brand spillover effect from blockchain technology is greater than the blockchain cost effect. In addition, when the blockchain-sensitive consumer market accounts for a relatively large market, no matter how the brand value effect with blockchain and the cost sensitivity to the effort are valued within their range, the profit level of the new e-tailer is always greater than that of the traditional e-tailer. Finally, e-tailers are always more profitable in the agency model regardless of the scope of the cost sensitivity to effort. That is, the agency model may be a more efficient way to improve the benefits of supply chain members.

7.3 Managerial insights

This section provides relevant managerial insights and hopes to help managers adopt blockchain technology more scientifically and optimize supply chain channels. Managerial insights can be summarized as follows:



https://doi.org/10.1371/journal.pone.0297978.g024

First, managers should be aware that the use of blockchain technology cannot unconditionally benefit the platform. The use of blockchain is not free, the cost involves technical maintenance and product information collection, put information on the chain and update information, etc. Therefore, these costs due to the blockchain cannot be ignored by managers. Specifically, if the cost sensitivity to effort is high, managers should reduce their effort as much as possible to obtain greater profits. In contrast, if the cost sensitivity to effort is low, they should increase their effort to obtain more benefits.

Second, after incorporating the cost sensitivity to effort into consideration of the company's operating cost, it was analysed together with relevant influencing factors. We find that the cost of using blockchain is closely related to the brand value effect it brings and the market share of blockchain-sensitive consumers. Furthermore, managers can use blockchain technology in the following conditions: the blockchain-sensitive consumer market accounts for a relatively large proportion, or although the blockchain-sensitive consumer market accounts for a relatively small proportion, the brand value effect with blockchain is relatively large and the cost sensitivity to the effort is small. Only under these conditions can blockchain technology help improve the company's profit level.

Third, from another perspective, not using blockchain technology does not mean that companies will fall behind. Whether the use of blockchain can improve the company's profit level depends on the combined effect of many factors. Companies that do not use blockchain technology can improve channel sales and increase profitability through marketing efforts. For example, improving customer experience before purchasing products, strengthening after-



https://doi.org/10.1371/journal.pone.0297978.g025

sales service, etc. Therefore, managers should evaluate specific business conditions and make scientific decisions under the premise of fully considering the combined effects of multiple factors on whether to use blockchain technology.

Finally, in the selection of sales scenarios, managers should choose more efficient scenarios to maximize profitability. Given this problem, the agency model may be a more efficient way to improve the benefits of supply chain members under certain conditions.

8. Conclusion

The research in this article shows that the use of blockchain technology is indeed a means to help companies gain a dominant position in a fiercely competitive environment, but whether e-platform retailers should blindly follow the trend and adopt this technology is worthy of indepth discussion in a dual-channel supply chain. Our results show that in a dual-channel supply chain, blockchain technology can only provide value under certain specific conditions. This discovery provides a good reference for the development of dual-channel supply chains and the scientific application of blockchain technology, helping channel participants make better decisions for their retail channel.

However, this paper still has some limitations. Thus, there are several interesting directions for the future that deserve further expansion based on this paper. First, this article is based on the horizontal competition of e-tailers in a dual-channel supply chain in the context of block-chain. In fact, there are also some situations in which upstream suppliers implement



https://doi.org/10.1371/journal.pone.0297978.g026

blockchain technology. Therefore, incorporating upstream suppliers into the dual-channel supply chain to consider the issue of vertical competition with downstream retailers will be a further research direction. Second, while we find that the agency scenario in the extended model may be a more efficient choice, we do not optimize supply chain member profits from a vertical coordination perspective. Accordingly, developing a win–win coordination mechanism to achieve a profit-Pareto-improving situation among platforms, supplies and retailers may be our next point. Third, this paper regards retail price as an exogenous variable without considering its impact on basic demand. Therefore, we will introduce the retail price into the

Table 7.	Summary	of the	findings.
----------	---------	--------	-----------

Findings	Logical proof	Explanation
The impact of blockchain technology adoption costs on companies	$\uparrow(\sqrt{-rac{B_3}{A_3}}\leqlpha<1),\Pi_{\mathcal{S}}^*\uparrow;$	• If the cost sensitivity to effort is high, managers should reduce their effort as much as possible to obtain greater profits.
	$ \begin{array}{l} \uparrow \ (-\frac{B_6}{A_6} \leq \alpha < 1), \ \Pi^{B*}_{ER_1} \ \uparrow; \\ \uparrow \ (-\frac{B_{11}}{A_{11}} \leq \alpha < 1), \ \Pi^{E*}_{ER_2} \ \uparrow. \end{array} $	• If the cost sensitivity to effort is low, they should increase their effort to obtain more benefits.
Conditions of use of blockchain technology	$egin{array}{l} rac{1}{2} \leq heta < 1, ext{ or if } 0 < heta < rac{1}{2} ext{ and } \ \gamma_b \geq rac{2ap(1-eta)(1-2 heta)}{(5-2eta)(eta+2)}, 0 < lpha < -rac{B_{16}}{A_{16}}. \end{array}$	 The blockchain-sensitive consumer market accounts for a relatively large proportion, managers should adopt blockchain. The blockchain-sensitive consumer market accounts for a relatively small proportion, the brand value effect with blockchain is relatively large and the cost sensitivity to the effort is small, managers should adopt blockchain.
What would be a better model to adopt	$\frac{\frac{\partial \Pi_{k_{1}}^{A*}}{\partial \chi} > \frac{\partial \Pi_{ER_{1}}^{B*}}{\partial \chi};}{\frac{\partial \Pi_{ER_{2}}^{A*}}{\partial \chi} > \frac{\partial \Pi_{ER_{2}}^{E*}}{\partial \chi}.$	• When the cost sensitivity to effort is low enough, the agency model may be a more efficient way to improve the benefits of supply chain members under certain conditions.

https://doi.org/10.1371/journal.pone.0297978.t007

model as an endogenous variable to further study the impact of retail prices on the profits of all supply chain members in future work. Finally, we focus our analysis on the traceability of blockchain, while other ingredients, such as smart contracts and accurate delivery, can be further considered to investigate the blockchain's advantages. Future research can examine their relationship together.

Supporting information

S1 Appendix. (DOCX)

Author Contributions

Conceptualization: Qian Chen, Dan Yang, Sen Liu.

Data curation: Xuejian Yang.

Formal analysis: Qian Chen, Xuejian Yang, Chuchu Liao.

Funding acquisition: Xuejian Yang, Sen Liu.

Investigation: Qian Chen, Xuejian Yang, Sen Liu, Chuchu Liao.

Methodology: Qian Chen, Xuejian Yang.

Project administration: Sen Liu.

Resources: Xuejian Yang, Dan Yang, Sen Liu, Chuchu Liao.

Software: Dan Yang, Chuchu Liao.

Supervision: Sen Liu.

Validation: Dan Yang.

Visualization: Dan Yang, Chuchu Liao.

Writing - original draft: Qian Chen.

Writing - review & editing: Sen Liu.

References

- 1. Ivanov D, Dolgui A, Sokolov B. The impact of digital technology and Industry 4.0 on the ripple effect and supply chain risk analytics. Int J Prod Res. 2019; 57(3):829–46.
- Banerjee M, Lee J, Choo K-KR. A blockchain future for internet of things security: a position paper. Digital Commun Networks. 2018; 4(3):149–60.
- Hassan MU, Rehmani MH, Chen J. Privacy preservation in blockchain based IoT systems: Integration issues, prospects, challenges, and future research directions. Future Gener Comput Syst. 2019; 97:512–29.
- Li Z, Wang WM, Liu G, Liu LN, He JD, Huang GQ. Toward open manufacturing A cross-enterprises knowledge and services exchange framework based on blockchain and edge computing. Ind Manage Data Syst. 2018; 118(1):303–20. https://doi.org/10.1108/imds-04-2017-0142 WOS:000423366900015.
- Jiang R, Kang Y, Liu Y, Liang Z, Duan Y, Sun Y, et al. A trust transitivity model of small and mediumsized manufacturing enterprises under blockchain-based supply chain finance. Int J Prod Econ. 2022; 247:108469. https://doi.org/10.1016/j.ijpe.2022.108469.
- Jiang R, Xin Y, Chen Z, Zhang Y. A medical big data access control model based on fuzzy trust prediction and regression analysis. Appl Soft Comput. 2022; 117:108423. <u>https://doi.org/10.1016/j.asoc.2022. 108423</u>.
- Kamath R. Food traceability on blockchain: Walmart's pork and mango pilots with IBM. JBBA. 2018; 1 (1):3712.

- Dai B, Nu Y, Xie X, Li J. Interactions of traceability and reliability optimization in a competitive supply chain with product recall. Eur J Oper Res. 2021; 290(1):116–31.
- 9. Jiang R, Han S, Yu Y, Ding W. An access control model for medical big data based on clustering and risk. Information Sciences. 2023; 621:691–707. https://doi.org/10.1016/j.ins.2022.11.102.
- Easley D, O'Hara M, Basu S. From mining to markets: The evolution of bitcoin transaction fees. J Financ Econ. 2019; 134(1):91–109.
- 11. Ilk N, Shang G, Fan S, Zhao JL. Stability of Transaction Fees in Bitcoin: A Supply and Demand Perspective. MIS Quart. 2021; 45(2).
- Cole R, Stevenson M, Aitken J. Blockchain technology: implications for operations and supply chain management. Supply Chain Manag. 2019; 24(4):469–83. https://doi.org/10.1108/scm-09-2018-0309 WOS:000479226300002.
- 13. De Giovanni P. Blockchain and smart contracts in supply chain management: A game theoretic model. Int J Prod Econ. 2020; 228:107855.
- 14. Chang JA, Katehakis MN, Shi JJ, Yan Z. Blockchain-empowered Newsvendor optimization. Int J Prod Econ. 2021; 238:108144.
- Saberi S, Kouhizadeh M, Sarkis J. Blockchain technology: a panacea or pariah for resources conservation and recycling? Resour Conserv Recy. 2018; 130(March):80–1.
- Dolgui A, Ivanov D, Potryasaev S, Sokolov B, Ivanova M, Werner F. Blockchain-oriented dynamic modelling of smart contract design and execution in the supply chain. Int J Prod Res. 2020; 58(7):2184– 99.
- Song H, Vajdi A, Wang Y Zhou J. Blockchain for consortium: a practical paradigm in agricultural supply chain system. Expert Syst Appl. 2021; 184:115425.
- Qiao Y, Lan Q, Zhou Z, Ma C. Privacy-preserving credit evaluation system based on blockchain. Expert Syst Appl. 2022; 188:115989.
- Sunny J, Undralla N, Pillai VM. Supply chain transparency through blockchain-based traceability: An overview with demonstration. Comput Ind Eng. 2020; 150:106895.
- Bumblauskas D, Mann A, Dugan B, Rittmer J. A blockchain use case in food distribution: Do you know where your food has been? Int J Inf Manage. 2020; 52:102008.
- Hastig GM, Sodhi MS. Blockchain for supply chain traceability: Business requirements and critical success factors. Prod Oper Manag. 2020; 29(4):935–54.
- Pournader M, Shi Y, Seuring S, Koh SL. Blockchain applications in supply chains, transport and logistics: a systematic review of the literature. Int J Prod Res. 2020; 58(7):2063–81.
- Mukherjee AA, Singh RK, Mishra R, Bag S. Application of blockchain technology for sustainability development in agricultural supply chain: justification framework. Oper Manage Res. 2021:1–16.
- Liu P, Long Y, Song H-C, He Y-D. Investment decision and coordination of green agri-food supply chain considering information service based on blockchain and big data. J Cleaner Prod. 2020; 277:123646.
- Liu RH, Tan CQ, Zhao CW. Pricing and coordination of vaccine supply chain based on blockchain technology. Internet Res. 2021; 31(6):2096–119. <u>https://doi.org/10.1108/intr-09-2020-0508</u> WOS:000677661500001.
- Zhang Q, Jiang X, Zheng Y. Blockchain adoption and gray markets in a global supply chain. Omega. 2023; 115:102785.
- Xu X, Yan L, Choi T-M, Cheng T. When is it wise to use blockchain for platform operations with remanufacturing? Eur J Oper Res. 2023; 309(3):1073–90.
- Wu X-Y, Fan Z-P, Cao B-B. An analysis of strategies for adopting blockchain technology in the fresh product supply chain. Int J Prod Res. 2023; 61(11):3717–34.
- 29. Zhang T, Dong P, Chen X, Gong Y. The impacts of blockchain adoption on a dual-channel supply chain with risk-averse members. Omega. 2023; 114:102747.
- Zhang T, Li P, Wang N. Multi-period price competition of blockchain-technology-supported and traditional platforms under network effect. Int J Prod Res. 2023; 61(11):3829–43.
- Kouhizadeh M, Saberi S, Sarkis J. Blockchain technology and the sustainable supply chain: Theoretically exploring adoption barriers. Int J Prod Econ. 2021; 231:107831.
- Moosavi J, Fathollahi-Fard AM, Dulebenets MA. Supply chain disruption during the COVID-19 pandemic: Recognizing potential disruption management strategies. Int J Disaster Risk Reduct. 2022; 75:102983. https://doi.org/10.1016/j.ijdrr.2022.102983 PMID: 35475018
- Simonetto M, Sgarbossa F, Battini D, Govindan K. Closed loop supply chains 4.0: From risks to benefits through advanced technologies. A literature review and research agenda. Int J Prod Econ. 2022:108582.

- 34. Fathollahi-Fard AM, Dulebenets MA, Hajiaghaei–Keshteli M, Tavakkoli-Moghaddam R, Safaeian M, Mirzahosseinian H. Two hybrid meta-heuristic algorithms for a dual-channel closed-loop supply chain network design problem in the tire industry under uncertainty. Adv Eng Inf. 2021; 50:101418.
- Asghari M, Afshari H, Mirzapour Al-e-hashem S, Fathollahi-Fard AM, Dulebenets MA. Pricing and advertising decisions in a direct-sales closed-loop supply chain. Comput Ind Eng. 2022; 171:108439.
- Berlin D, Feldmann A, Nuur C. The relatedness of open-and closed-loop supply chains in the context of the circular economy; framing a continuum. Cleaner Logistics and Supply Chain. 2022; 4:100048.
- Balakrishnan A, Sundaresan S, Zhang B. Browse-and-switch: Retail-online competition under value uncertainty. Prod Oper Manag. 2014; 23(7):1129–45.
- Gao F, Su X. Omnichannel retail operations with buy-online-and-pick-up-in-store. Manage Sci. 2017; 63(8):2478–92.
- Colombo S, Matsushima N. Competition between offline and online retailers with heterogeneous customers. Rev Ind Organ. 2020; 57(3):647–64.
- Chai L, Wu DD, Dolgui A, Duan Y. Pricing strategy for B&M store in a dual-channel supply chain based on hotelling model. Int J Prod Res. 2021; 59(18):5578–91.
- Abhishek V, Jerath K, Zhang ZJ. Agency selling or reselling? Channel structures in electronic retailing. Manage Sci. 2016; 62(8):2259–80.
- 42. Choi T-M, Feng L, Li R. Information disclosure structure in supply chains with rental service platforms in the blockchain technology era. Int J Prod Econ. 2020; 221:107473.
- Jiang S, Li Y, Wang S, Zhao L. Blockchain competition: The tradeoff between platform stability and efficiency. Eur J Oper Res. 2022; 296(3):1084–97.
- 44. Shi H, Liu Y, Petruzzi NC. Consumer heterogeneity, product quality, and distribution channels. Manage Sci. 2013; 59(5):1162–76.
- Herbon A. Optimal pricing and replenishment of an expiring inventoried product under heterogeneous consumer sensitivities. Decision Sci. 2018; 49(3):522–52.
- Kopczewski T, Sobolewski M, Miernik I. Bundling or unbundling? Integrated simulation model of optimal pricing strategies. Int J Prod Econ. 2018; 204:328–45.
- Homburg C, Lauer K, Vomberg A. The multichannel pricing dilemma: Do consumers accept higher offline than online prices? Int J Res Mark. 2019; 36(4):597–612.
- Lin J, Zhang J, Cheng TCE. Optimal pricing and return policy and the value of freight insurance for a retailer facing heterogeneous consumers with uncertain product values. Int J Prod Econ. 2020; 229:107767.
- 49. Lei Y, Liu Q, Shum S. Warranty pricing with consumer learning. Eur J Oper Res. 2017; 263(2):596–610.
- Zheng R, Wang R, Yang C. How consumer valuation heterogeneity impacts firms' profit: Peer influence makes a difference. Comput Ind Eng. 2020; 149:106846.
- Qiu R, Li C, Sun M. Impacts of consumer virtual showrooming behavior on manufacturer and retailer strategic decisions in a dual-channel supply chain. Eur J Oper Res. 2024; 313(1):325–42.
- Wang R, Wang Q, Chiang W-yK. Optimal promotional mix and pricing when faced with uncertain product value. Eur J Oper Res. 2024; 313(2):637–51.
- Li R, Basu AK. Pricing Strategy for GM Food: Impact of Consumer Attitude Heterogeneity and GMO Food Labelling. Ann Oper Res. 2020; 291(1):463–74.
- 54. Meng Q, Wang Y, Zhang Z, He Y. Supply chain green innovation subsidy strategy considering consumer heterogeneity. J Cleaner Prod. 2021; 281:125199.
- Sharma P, Srivastava A, Sharma V, Singh N, Nijjer S. Understanding consumer repurchase intentions towards luxury retail brands: Evidence from an emerging market. Journal of Retailing and Consumer Services. 2024; 76:103570.
- Li Y, Jiang S, Shi J, Wei Y. Pricing strategies for blockchain payment service under customer heterogeneity. Int J Prod Econ. 2021; 242:108282.
- Pun H, Swaminathan JM, Hou P. Blockchain adoption for combating deceptive counterfeits. Prod Oper Manag. 2021; 30(4):864–82.
- van Hoek R. Exploring blockchain implementation in the supply chain: Learning from pioneers and RFID research. Int J Oper Prod Man. 2019; 39(6/7/8):829–59. <u>https://doi.org/10.1108/ijopm-01-2019-0022</u> WOS:000497201400002.
- Genc TS, De Giovanni P. Closed-loop supply chain games with innovation-led lean programs and sustainability. Int J Prod Econ. 2020; 219:440–56.